

Overcoming the Carbon Trap: Climate Policy and Technology Tipping

Working Paper

Author(s): Bretscher, Lucas; Komarov, Evgenij; Leuthard, Matthias

Publication date: 2023-11

Permanent link: https://doi.org/10.3929/ethz-b-000644504

Rights / license: In Copyright - Non-Commercial Use Permitted

Originally published in: Economics Working Paper Series 23/386

Funding acknowledgement: 870245 - Global Excellence in mOdeling Climate and Energy Policies (EC)



CER-ETH – Center of Economic Research at ETH Zurich

Overcoming the Carbon Trap: Climate Policy and Technology Tipping

L. Bretschger, E. Komarov, M. Leuthard

Working Paper 23/386 November 2023

Economics Working Paper Series



Eidgenössische Technische Hochschule Zürich Swiss Federal Institute of Technology Zurich

Overcoming the carbon trap: Climate policy and technology tipping^{*}

Lucas Bretschger[†] Evgenij Komarov[‡] Matthias Leuthard[§]

This Version: November 29, 2023

Abstract

We construct an overlapping generations model in which the choice between dirty and clean technology hinges on the economy's capital stock, susceptible to climate-induced depreciation. The process of capital accumulation contributes to environmental emissions, yet their intensity can be mitigated through a shift to cleaner production methods. The tipping point of technological transition is endogenously determined, leading to a diverse range of potential long-term outcomes shaped by capital endowment, pollution intensity, climate vulnerability, and clean factor productivity. Our analysis reveals the possibility of an economy converging into a "carbon trap", characterized by a sustained equilibrium marked by elevated pollution and diminished income, despite the feasibility of pursuing green growth. Additionally, we present optimal policy measures and simulations that highlight the temporal disparities between the socially optimal timing for transitioning to green technology and the timing dictated by market forces. Finally, to account for the high upfront costs of starting clean production, we extend the model by including a non-convexity in the production structure of the clean technology.

Keywords: Carbon trap, technology tipping, climate damages, climate policy

JEL Classification:

^{*}We thank Ara Jo as well as the participants of the SURED 2022 conference in Ascona and the PSE-ETH workshop 2023 in Paris for the valuable comments and the fruitful discussions on the topic. This paper is part of a project that has received funding from the European Union's Horizon 2020 research and innovation program under the Marie Skłodowska-Curie grant agreement No 870245.

[†]CER-ETH – Center of Economic Research at ETH Zurich, lbretschger@ethz.ch

[‡]CER-ETH – Center of Economic Research at ETH Zurich, ekomarov@ethz.ch

[§]CER-ETH – Center of Economic Research at ETH Zurich, mleuthard@ethz.ch

1 Introduction

According to standard growth theory, economic development is driven by the accumulation of capital stocks. Capital refers to the productive capacities of an economy, such as machinery, roads, buildings, infrastructure or arable land. Recent years have shown that negative climate events, e.g. floods, hurricanes and landslides, destroy some of the existing capacities. As a consequence, climate change not only reduces current production, but also generates negative growth effects, as the destruction of capital and infrastructure reduces net capital formation (Dietz & Stern, 2015). Empirical evidence suggests that less developed countries are disproportionately more affected by these damages than industrialized countries, even though they are often the least responsible for the greenhouse gas emissions that drive climate change (IPCC, 2023). After a climate shock, production capacities need to be rebuilt, which takes time and requires a reduction in consumption to enable additional investments.

To curb global warming and avoid major economic damages from climate change, a rapid transition to fossil fuel-free technologies is of utmost importance. Since climate change is a global externality, the transformation of the energy system must take place not only in rich countries but also in the developing world. It is widely acknowledged that the transition to cleaner technologies requires a certain amount of production capacity in the form of physical capital.¹ As a consequence, the transition is becoming increasingly difficult or even infeasible for some climate-vulnerable and capital-constrained countries, as climate-related natural disasters adversely affect the existing capital stock — a potential development barrier that has been largely neglected by the existing literature. More specifically, if developing countries have to devote an increasing share of their scarce capital to replacing the existing capital stock, they will not be able to build the capital stock needed for technology transition and thus to generate long-term economic growth based on fossil-free technologies; even though clean and decentralized renewable energy technologies offer great potential to drive electrification and economic development especially in remote and poor rural areas without centralized energy access (Edenhofer et al., 2013).

Hence, there is growing concern that developing countries will be caught in an "environmental poverty trap" that is, a long-term equilibrium of low income and high pollution (Leichenko & Silva, 2014).² The experts of IPCC (2022) conclude with high con-

¹There is ample empirical evidence for this capital requirement, see e.g. Komen et al. (1997), Varvarigos (2014) and Charlier et al. (2022); and a large body of theoretical work relies on this assumption. Specifically, economic models in which a critical stock of physical capital must be accumulated before the economy can transition from dirty to clean technologies include, among others, Stokey (1998), Iwaisako (2002), Matsuyama (2007), Cunha-E-Sa & Reis (2007), Pommeret & Schubert (2009), Damsgaard (2012), Varvarigos (2014) and Charlier et al. (2022). This is also in the spirit of the Environmental Kuznets Curve (EKC) hypothesis, which posits an inverted U-shape relationship between per capita income and pollution (Grossman & Krueger, 1995).

 $^{^{2}}$ For instance, Bretschger & Valente (2011) use an endogenous growth model to show that climateinduced capital depreciation may lead to "climate poverty traps" especially in climate-vulnerable and

fidence that "impacts of climate change together with non-climatic drivers can create poverty–environment traps that may increase the probability of long-term and chronic poverty".

The relationship between capital accumulation, pollution and technology transition is complex, but a suitable model can reduce complexity to a small number of well-specified relationships and identify the factors that determine the transition to cleaner technologies. By developing and using such a tractable dynamic macroeconomic framework, we illustrate the dual role of capital accumulation for technology transition and long-term economic development: On the one hand, large-scale capital investments in infrastructure, machinery and other physical assets enable the transition to low-carbon technologies. On the other hand, installing capital today leads to pollution and thus climate change, which harms capital in the future and makes it increasingly difficult to maintain or increase capital levels, potentially delaying the point in time at which the economis that are unable to build the production capacities needed for technology transition may be trapped in a perpetual state of low income and high pollution, even if the clean technology is technologically available and sufficiently productive to generate long-term economic growth. We refer to this outcome as a carbon trap in the following.

To analyze the emergence and the possible avoidance of carbon traps, we construct a two-period overlapping generations (OLG) model in which the choice of technology is endogenous and capital investment is the driver of economic growth. With the goal of maximizing the return on investment, households can invest in either a dirty or a clean technology, which differ in terms of their capital–, learning– and pollution intensity. Since the returns are monotonous, it follows that the economy in equilibrium uses only one technology. Pollution is a negative externality of capital accumulation, and climate change causes damages to the stock of physical capital which constitutes the productive capacity of the economy. Pollution intensity of capital can be reduced by switching to cleaner production — but only after the economy has paid a fixed costs in terms of capital. Our framework allows each generation to reassess the return on investment and thus to revert to dirty production in any period, which may occur, for example, if the clean technology is not productive enough or if the necessary physical capital (e.g. in the form of fossil-free infrastructure) is lacking.³

When modelling the clean technology, we incorporate several stylized facts into our model. First, we take into account the fact that clean technologies tend to be more capital intensive than dirty technologies (IEA, 2022). Second, following the empirical literature

capital-poor countries. Other related contributions involving an "environmental poverty trap" are, e.g., Xepapadeas (1997), Palivos et al. (2010), Antoci et al. (2011), Fodha & Seegmuller (2014), Golub & Toman (2016), Dao & Edenhofer (2018) and Barbier & Hochard (2019).

³A recent example for such a technology shift is the switch from gas to more polluting coal in Germany (LeMonde, 2022).

on learning rates⁴ (Rubin et al., 2015) and knowledge spillovers (Dechezleprêtre et al., 2013), we consider the fact that newer and cleaner technologies tend to have relatively high learning– and spillover intensities compared to mature fossil-based technologies. Accordingly, we assume that the clean technology is subject to higher learning effects in the process of capital accumulation compared to the dirty technology. Third, since a large part of the existing capital stock is designed for fossil-based technologies, the transition to cleaner production leads to "stranded assets" (McKibben, 2012). Hence, we assume that the available productive capacity in the form of physical capital is reduced as the economy transitions from the dirty to the clean technology, which is captured by a fixed fraction of capital that is lost at the time of the transition. Indeed, during the energy transition, part of the capital such as the distribution infrastructure for electricity can be preserved, while another part, such as the coal plants for power generation, is lost. Put differently, changing to a clean technology such as solar panels renders the coal plants obsolete but continues to use the existing infrastructure for delivering energy to firms and households. Finally, in an extension, we realize that clean technologies typically require large upfront investments in capital infrastructure before they can become productive, see e.g. Nelson & Shrimali (2014).⁵ As a consequence, there exist huge non-convexities when building a fossil-free infrastructure e.g. in the form of charging stations for electric vehicles. To account for this feature, we extend the baseline model by incorporating a non-convexity in the production structure of the clean technology in the form of a minimum threshold for capital.

In this framework, we find that the economy's technology choice depends on an endogenous threshold for capital: Economies whose capital stock exceeds this capital threshold — henceforth denoted as tipping point — invest in the clean technology, while it is optimal for economies below this tipping point to invest in the dirty technology.⁶ This results in a rich set of possible long-run dynamics for the economy depending on the interaction of two conditions, the "transition condition" and the "green growth condition". The transition condition establishes a threshold for clean factor productivity above which a transition to clean technologies can occur before the economy converges to the non-growing polluting steady-state. The green growth condition provides a threshold for clean factor productivity above which the economy can achieve long-run growth with the clean technology. The actual growth pattern of the economy is then determined by the TFP value of the clean technology and it's initial capital endowment.

 $^{^{4}}$ As e.g. documented by McDonald & Schrattenholzer (2001) and Rubin et al. (2015), the cost of renewable technologies decreases with cumulative installed capacity at a stable rate, and no comparable effect exists for dirty, mature technologies.

⁵The upfront capital costs for wind, solar, and hydro energy are 84-93% of the total project costs while for coal and gas they amount to only 24-37% (Nelson & Shrimali, 2014).

 $^{^{6}}$ We use the term tipping point to indicate that the stage of development can trigger a sudden and significant shift in the adoption or dominance of a particular technology, see e.g. Otto et al. (2020) for a more detailed discussion.

Our model shows that capital-poor economies that possess a low clean factor productivity converge to a non-growing dirty steady state, even though a steady state with sustained green growth would be achievable, i.e., a carbon trap arises. This trap does not occur with a high TFP value for the clean technology. If an economy possesses a particularly low clean factor productivity, the steady state with positive growth vanishes and the economy converges to the dirty steady state regardless of its initial capital endowment. Because less developed economies tend to have a low clean factor productivity and capital endowment, and are more exposed to the impacts of climate change, they are less likely to build the capital stock needed for technology transition, and are therefore particularly at risk of being caught in a carbon trap or in a dirty steady state, where economic growth necessarily comes to an end.

The introduction of a minimum capital requirement for the clean technology represents an additional barrier to sustained green growth and expands the set of possible long-term outcomes for the economy. In particular, we show that in the presence of a capital threshold, a carbon trap can arise even when the economy has a relatively high clean factor productivity. Moreover, our model reveals that in the presence of a capital threshold, the economy can be caught in a carbon trap even after the transition to clean production.

We further derive the socially optimal tipping point which internalizes all non-market effects related to climate change, learning, and transition costs in switching from dirty to clean capital. Using standard parameter values, we find that this tipping point is lower than the tipping point that arises in the market economy. To determine the difference in transition times between the market outcome and the social optimum, we perform numerical simulations and show that the socially optimal tipping point occurs three generations or, equivalently, about 90 years before the market tipping point. We then analyze policies to promote a timely transition to clean production. We show both theoretically and numerically that a tax on dirty production or a subsidy for clean production can induce the socially optimal tipping point in the market economy, reducing the risk of falling into a permanent state of low income and high pollution.

Our analysis is related to different strands of literature. First, we build on the aforementioned work of Bretschger & Valente (2011), which introduces climate-induced capital depreciation into an endogenous growth model and shows that climate change can lead to negative consumption growth and stagnation traps due to its impact on capital depreciation, especially in climate-vulnerable and less developed economies.⁷ While we use an OLG model instead of a continuous time model, we extend their analysis by including a discrete technology choice that allows the economy to escape stagnation by switching to cleaner production.

⁷Other related papers in which climate change damages the stock of physical capital include Bretschger & Suphaphiphat (2014), Bretschger (2017), and Bretschger & Karydas (2019).

Second, we rely on OLG models in which the choice of technology is endogenous and dependent on a critical stock of capital, as in Iwaisako (2002) and Matsuyama (2007), and with Asano et al. (2021) and Asano et al. (2022) being more recent contributions.⁸ We recast this analysis in an environmental context where technologies differ in terms of their pollution intensity, thereby damaging the stock of physical capital to varying degrees.

Third, we also contribute to the analysis of the optimal timing for the adoption of clean technologies. In this context, the optimal capital stock and timing of technology transition is endogenously determined in Cunha-E-Sa & Reis (2007), Pommeret & Schubert (2009) and Damsgaard (2012), and exogenously specified in Charlier et al. (2022). The latter presents economic development as a process of structural change and demonstrates for different policy schemes that it is always optimal to reach the capital stock that coincides with structural change as quickly as possible. Cunha-E-Sa & Reis (2007) analyze the optimal timing for adopting a greener technology when that technology offers higher productivity but entails some adjustment costs, and find that the optimal timing depends on the marginal utility of environmental quality with respect to consumption; Pommeret & Schubert (2009) introduce uncertainty in this framework and find that higher uncertainty leads to an earlier adoption of the green technology. Damsgaard (2012) shows that the optimal timing for the introduction of a green technology depends on the size of the capital stock relative to the non-renewable resource stock, with a small capital stock leading to a delayed adoption of the alternative fossil-free technology.

We extend this strand of literature in several ways. First, the presented papers omit the negative impacts of climate change on growth. For instance, they assume that climate change affects utility (e.g. Cunha-E-Sa & Reis (2007), Pommeret & Schubert (2009) and Charlier et al. (2022)) or abstract completely from climate change impacts (e.g. Damsgaard (2012)). However, by destroying part of the existing production capacities, global warming has a significant and long-lasting impact on the growth prospects of economies and thus on their choice of technologies. Second, in addition to the existing literature, we numerically determine the difference in transition times between the market outcome and the social optimum.⁹ Third, we extend the analysis from a constellation in which the economy can change technology only once to a model with overlapping generations, in which it may be optimal for agents to revert to dirty production in any period, depending on the evolution of the investment returns of the technologies.¹⁰

⁸Other related contributions are Narita (2010), Asano et al. (2012) and Umezuki & Yokoo (2019).

 $^{^{9}}$ In this respect, closest to our contribution is Pommeret & Schubert (2009), who also derive the capital stock needed for technology transition for both the market economy and the social optimum, but they do not determine the difference in transition timings.

 $^{^{10}}$ In growth models with a representative infinite-lived household, there is no incentive to revert to the dirty technology once the transition costs for the technology change have been incurred, regardless of how the economy performs with the clean technology, see e.g. Cunha-E-Sa & Reis (2007) and Pommeret & Schubert (2009).

Finally, our work is closely related to the strand of literature on environmental poverty traps induced by the negative effects of pollution on life expectancy and the resulting reduced incentives for long-term capital accumulation, see Dugan et al. (2022) for a recent survey on this matter.¹¹ In this framework, Ikefuji & Horii (2007), Varvarigos (2014) and Dao & Edenhofer (2018) introduce a discrete technology choice into an overlapping generations model in which survival probability depends on both pollution and income, and multiple equilibria can emerge due to a virtuous cycle¹² of pollution, longevity, and investment.¹³ We abstract from the health channel and focus exclusively on the impact of capital formation on technology transition and show that carbon traps can arise even if we neglect the interplay between the environment, life expectancy, and capital accumulation.

The remainder of the paper is structured as follows: In Section 2 we present the baseline model and its dynamics. Section 3 provides the social optimum of the economy, and Section 4 discusses the impact of a carbon tax and a subsidy on market outcomes. The numerical simulations are performed in Section 5, and Section 6 extends the baseline model by introducing a non-convexity in the production structure of the clean technology in form of a minimal threshold for capital. Section 7 discusses the results and, finally, Section 8 concludes.

2 The Model

In this section, we present our model framework and motivate the assumptions, starting with the changing climate and then proceeding with the households, technologies, firms, and technology choices, in turn.

2.1 Climate change

We consider a discrete time OLG world economy, in which either a clean (c) or a dirty (d) technology can be employed for production. Total labor endowment \overline{L} is normalized to unity, so that aggregate and per-capita variables coincide. Pollution is a by-product of capital accumulation. The stock of pollution at instant t is denoted by P_t and increases proportionally with the stock of capital used in production, $k_{i,t}$, where the index $i \in \{c, d\}$ refers to the employed technology. The intensity of pollution varies depending on the used technology. For an economy that operates with the dirty technology, pollution evolves

 $^{^{11}}$ This strand builds on the seminal contributions of Chakravorty et al. (2008) and Mariani et al. (2010).

¹²This cycle underlies the following reasoning: Higher environmental quality extends life expectancy, which in turn increases the incentives to save and accumulate capital. This allows the economy to reach the level of capital that enables the transition to clean production, which in turn increases environmental quality and thus life expectancy.

¹³To generate multiple equilibria, Dao & Edenhofer (2018) introduce a nonlinear regeneration function for the environment in addition to the presented cycle.

according to

$$\Delta P_t = \varphi k_{d,t},\tag{1}$$

where $\varphi > 0$ is the constant pollution intensity of capital.¹⁴ With a switch from dirty to clean technology the economy can reduce its pollution intensity by a constant amount $\iota \in (0, 1]$, so that (1) changes to

$$\Delta P_t = \varphi(1-\iota)k_{c,t},\tag{2}$$

where $\iota = 1$ implies that a pollution-free technology is employed. Building on Bretschger & Valente (2011) and the references therein, we assume that climate change affects capital accumulation by increasing the rate of capital depreciation. In what follows, we assume that over the lifetime of a generation (i.e., over ≈ 30 years) capital can adapt to a higher pollution level without suffering from additional depreciation losses, so that capital destruction is determined by the change in pollution stock.¹⁵ We denote the amount of capital that is destroyed as a result of increased pollution by $D(\Delta P_t)$; it is added to the conventional depreciation rate (i.e., the rate of capital that is depreciated in the absence of climate change) measured by $\delta \in (0, 1)$. Assuming that the change in pollution stock impacts capital depreciation by a proportional factor η , we obtain

$$D(\Delta P_t) \equiv \eta \Delta P_t,$$

where $\eta > 0$ is a constant impact intensity parameter which represents the sensitivity of the capital stock to climate change. Depending on the technology in use, total depreciation amounts to

$$\delta k_{d,t} + D(\Delta P_t) = \delta k_{d,t} + \eta \Delta P_t = [\delta + \varphi \eta] \, k_{d,t} \equiv \Lambda_d k_{d,t}, \tag{3}$$

$$\delta k_{c,t} + D(\Delta P_t) = \delta k_{c,t} + \eta \Delta P_t = \left[\delta + (1-\iota)\varphi\eta\right]k_{c,t} \equiv \Lambda_c k_{c,t},\tag{4}$$

where Λ_i denotes the total rate of capital depreciation associated with the employment of technology $i \in \{d, c\}$, and is used below to simplify expressions. The size of capital depreciation depends on the pollution intensity of capital, which varies depending on the technology in use and the impact intensity of climate change.

¹⁴For simplicity, we abstract from a natural decay rate.

¹⁵A similar assumption is applied in Bretschger & Valente (2011), Bretschger & Suphaphiphat (2014) and Bretschger (2017). Considering the differences between the effects of the level and the rate of change of pollution, Tahvonen (1995) concludes that, especially in the context of climate change, damage also depends on the change of the pollution stock. Using an OLG model with a longer time horizon we restrict the analysis to the change of the stock pollutant assuming that capital after a generation is adapted to higher pollution levels.

2.2 Households

We assume a representative household with a two-period lifetime, so that at any point in time two generations coexist. An agent is labeled "young" in the first period and "old" in the second period. Young agents are endowed with one unit of labor which they supply inelastically to firms in exchange for wage income $w_{i,t}$. Capital is owned by old agents, who bequeath the capital stock net of depreciation to the young generation at the end of each period.¹⁶ The amount of capital inherited depends on the size of climate-related capital losses and thus on the technology choice of the old. Because they make their investment decisions solely with the goal of maximizing returns and without taking climate-related externalities into account, their choice of technology constitutes an intergenerational externality that affects the wealth of the young and thus their ability to save.

A young agent decides how much to consume (c_t^1) and save (s_t) out of total wealth, which consists of the inherited capital stock net of depreciation $(1 - \Lambda_i)k_t$ and wage income $w_{i,t}$. An old agent simply consumes its savings, where the return on investment for technology $i \in (c, d)$ in period t + 1 is denoted by $r_{i,t+1}$ and is specified in more detail in Section 2.5. Further, we introduce transition costs expressed by $\theta \in (0, 1)$, which reflect that part of the capital stock becomes obsolete when capital moves from dirty to clean technology, see Cunha-E-Sa & Reis (2007) and Pommeret & Schubert (2009) for a similar treatment.¹⁷ For simplicity, we assume that the reverse change, i.e., from clean to dirty, does not involve such costs.¹⁸ From that, we obtain the following budget constraints for an agent when being young and old, respectively:

$$w_{i,t} + (1 - \theta \psi - \Lambda_i)k_t = c_t^1 + s_t$$
, and
 $c_{t+1}^2 = r_{i,t+1}s_t$,

with $\theta \psi + \Lambda_i \leq 1$ and ψ being an indicator variable which accounts for the fixed costs in terms of capital; it takes the value of either zero or one, depending on whether or not a

¹⁶The reason for this may be altruistic motives or uncertainty about life expectancy, see e.g. Lines (2001) for a discussion. A similar assumption is used, e.g., in Cremers (2006), Dam (2006) and Karp et al. (2021). Alternatively, we could assume that the old household bequeaths only the share $q(1 - \Lambda_i)k_t$ with $q \in [0, 1]$ to the young generation and consumes the remaining share $(1 - q)(1 - \Lambda_i)k_t$ when they are old. However, this would not change our main results. For simplicity, we assume that the old generation bequeaths the entire capital stock to the young agent at the end of each period, i.e., q = 1.

¹⁷As outlined in the introduction, the part of the capital stock that becomes obsolete due to the switch in technology can be interpreted as "stranded" capital.

¹⁸More generally, we posit that switching back from clean to dirty energy generation methods is much less costly because old power plants can be recommissioned and do not have to be built from scratch.

technology change from dirty to clean occurs in a given period t:

$$\psi = \begin{cases} 1 & \text{if } t = t^{TP}, \\ 0 & \text{if } t \neq t^{TP}, \end{cases}$$

where $t = t^{TP}$ refers to the point in time at which the technology transition from dirty to clean occurs. By isolating s_t , the two period budget constraints can be consolidated to the following intertemporal budget constraint

$$w_{i,t} + (1 - \theta \psi - \Lambda_i)k_t = c_t^1 + \frac{c_{t+1}^2}{r_{i,t+1}}.$$
(5)

Moreover, lifetime utility of an individual born in t is specified as

$$U_t(c_t^1, c_{t+1}^2) = \log(c_t^1) + \beta \log(c_{t+1}^2),$$
(6)

where $\beta \in (0, 1)$ is the discount factor. Individuals maximize lifetime utility (6) subject to their intertemporal budget constraint (5), which leads to the familiar Euler equation:

$$c_{t+1}^2 = \beta r_{i,t+1} c_t^1.$$
(7)

By combining the Euler equation (7) with the intertemporal budget constraint (5), we obtain the following optimal saving function:

$$s_t = \frac{\beta}{1+\beta} \left[w_{i,t} + (1-\theta\psi - \Lambda_i)k_t \right], \quad i \in \{c, d\},$$
(8)

which is independent of $r_{i,t+1}$ due to the logarithm utility and states that the household saves a constant share of total wealth. We note that the young household receives income only from supplying labor while the old household receives income only from renting out capital. Hence, we consider the young as workers and the old as investors.

2.3 Technologies

There is one good in the economy whose output we denote by Y. This good is used for consumption and investment and is produced with either the clean or the dirty technology, with the respective outputs being Y_c and Y_d . There are infinitely many identical firms that are all price-takers and the production function for firm j that operates with either the clean or dirty technology reads

$$Y_{i,j} = \tilde{A}_i K_{i,j}^{\alpha_i} L_{i,j}^{1-\alpha_i}, \quad i \in \{c, d\},$$

where K_i is capital, L_i denotes labor and $\alpha_i \in (0, 1)$ refers to the capital intensity in production. Capital accumulation is assumed to have a positive effect on total factor productivity through learning-by-doing as in Romer (1986). Specifically, \tilde{A}_i is a measure of learning-by-doing effects and depends on the average capital ratio in production:

$$\tilde{A}_i = A_i k_i^{\nu_i}, \quad i \in \{c, d\},$$

where $\nu_i \geq 0$ and $A_i > 0$ are technology-specific parameters for the strength of the learning-by-doing effect and total factor productivity, respectively. Each firms is small compared to the total market and takes k_i as given. In particular, if $\nu_i > 0$ then positive externalities exist in production and if $\nu_i = 0$ learning effects are absent.¹⁹ Assuming that all firms are identical, we can express the production function for a firm operating with technology *i* as:

$$y_i = \tilde{A}_i k_i^{\alpha_i},$$

while actual output is determined by

$$y_i = A_i k_i^{\alpha_i + \nu_i}.$$

For labor, we assume perfect mobility, while for capital the previously established transition costs apply. Moreover, we assume in the following that the clean technology lags relatively behind the dirty technology in terms of total factor productivity, i.e., $A_c < A_d$.

2.4 Firms

The factor markets are perfectly competitive so that firms take prices for labor and capital as given. All prices are normalized using the final good as numeraire. Neglecting its individual impact on climate change and capital accumulation, a representative firm operating with either the dirty or clean technology maximizes its profit in every period by choosing how much labor and capital it employs. Since labor supply is equal to one and inelastic, we can express the factor prices as functions of the capital stock:

$$p_i(k_i) = \alpha_i A_i k_i^{\alpha_i + \nu_i - 1}, \tag{9}$$

$$w_i(k_i) = (1 - \alpha_i) A_i k_i^{\alpha_i + \nu_i},\tag{10}$$

where w_i and p_i stand for the wage and the rental price of capital for technology $i \in \{c, d\}$, respectively. As a result, we get the familiar optimality conditions, that is, the marginal product of capital corresponds to the rental price of capital and the marginal product of

¹⁹For simplicity, we abstract from negative externalities in production.

labor is equal to the wage. The shape of the marginal productivity of capital depends on the value of $\nu_i + \alpha_i$ and, for simplicity, we will write $\gamma_i = \alpha_i + \nu_i$. An overview of the factor prices for both technologies is provided in Table 1. Throughout this paper,

Technology:	Rental rate (p) :	Wage (w) :
Dirty technology:	$p_d = \alpha_d A_d k_d^{\gamma_d - 1}$	$w_d = (1 - \alpha_d) A_d k_d^{\gamma_d}$
Clean technology:	$p_c = \alpha_c A_c k_c^{\gamma_c - 1}$	$w_c = (1 - \alpha_c) A_c k_c^{\gamma_c}$

Table 1: Factor prices for both technologies

we assume that $\gamma_c > \gamma_d$ which is based on two empirical observations: First, empirical evidence argues in favor of higher learning rates for clean technologies compared to dirty ones (Rubin et al., 2015). Accordingly, we assume that the learning intensity in capital formation is higher for the clean as compared to the dirty technology, i.e., $\nu_c > \nu_d$. Second, capital intensity tends to be higher for clean as compared to dirty technologies (IEA, 2022), which implies that $\alpha_c \geq \alpha_d$, so that overall $\gamma_c > \gamma_d$.

2.5 Tipping points and the choice of technology

The household invests the entire capital stock in either the clean or the dirty technology, depending on which yields a higher return. Recall that a technology transition from dirty to clean involves transition costs (i.e., capital used in dirty production cannot be fully deployed in clean production), so that at the time of transition we have $k_c = (1-\theta)k_d$ with $\theta \in (0, 1)$. Since switching from clean to dirty technology does not involve such costs, the household's investment decision depends on which technology was used in the previous period. Hence, we distinguish four different cases of transitions and thus four different investment returns, where we denote the investment return for technology $i \in \{c, d\}$ by r_i . Using a standard definition of a return, r_i is specified as

$$r_i(k_{i,t}) = \frac{p_i(k_{i,t}) \cdot k_{i,t}}{k_t}, \quad i \in \{c, d\},$$

where $p_i(k_{i,t})$ is the rental price of capital and k_t stands for the aggregate capital stock regardless of the technology for which it is used. We derive the returns in all four cases in the Appendix and focus for now on the case where the economy transitions from dirty to clean technology.

Given these returns, the household chooses in each period the technology whose investment return is higher than the alternative, without taking into account climaterelated externalities. Thus, the investment problem of the household reads

$$\max_{i \in \{c,d\}} r_i(k_i).$$

Next, we determine the aggregate capital stock for which the household is indifferent between both technologies. Given a technology change from dirty to clean, the appropriate non-arbitrage condition reads $r_d(k) = r_c(k)$ or, equivalently,

$$p_d(k) = p_c[(1-\theta)k] \cdot (1-\theta).$$

Using the expressions for p_d and p_c from Table 1 yields

$$\alpha_d A_d k^{\gamma_d - 1} = \alpha_c A_c \left[(1 - \theta) k \right]^{\gamma_c - 1} \cdot (1 - \theta).$$

Note that the returns in our economy are monotonous, so the economy uses only one technology in equilibrium. Finally, solving for k yields

$$\bar{k}^{TP} \equiv k = \left(\frac{\alpha_d A_d}{\alpha_c A_c (1-\theta)^{\gamma_c}}\right)^{\frac{1}{\gamma_c - \gamma_d}},\tag{11}$$

with $\gamma_c > \gamma_d$ and thus $\bar{k}^{TP} \ge 0$. This critical value of k represents a tipping point at which the technology change from dirty to clean occurs. Accordingly, we refer to \bar{k}^{TP} as the first tipping point in the economy. If k_t is above this tipping point, then the return with the clean technology exceeds the one with the dirty technology and vice versa if k_t lies below. Hence, (11) represents the minimum level of capital above which it is optimal for the household to invest in the clean instead of the dirty technology.

We find that this tipping point increases with the cost to switch (high θ) and decreases with the distance in the learning intensities (high $\gamma_c - \gamma_d$) between the two technologies. The higher the strength of the learning-by-doing effect with the clean as compared to the dirty technology, the more rewarding an early transition to clean production and thus the lower the tipping point. Beside that, in line with intuition, high values of A_c and α_c and low values of A_d and α_d promote a technology switch in favor of the clean technology by reducing \bar{k}^{TP} .

Analogously, we can study the change from clean to dirty. Repeating the same steps, a second tipping point can be derived, which is given by

$$\underline{k}^{TP} \equiv k = \left(\frac{\alpha_d A_d}{\alpha_c A_c}\right)^{\frac{1}{\gamma_c - \gamma_d}}.$$
(12)

This tipping point provides the level of aggregate capital below which it is optimal for the household to invest in the dirty instead of the clean technology. Note that this second tipping point is smaller than the first one, i.e., $\underline{k}^{TP} < \overline{k}^{TP}$, as the costs θ play no role.

We summarize our findings in the following proposition:

Proposition 1. Depending on the economy's technology choice in period t-1 (its initial technology choice), the following tipping points arise in the economy in period t:

- i. Given that in t-1 the economy invested in the dirty technology, then in t the economy invests in the clean technology if $k_t \geq \bar{k}^{TP}$.
- ii. Given that in t-1 the economy invested in the clean technology, then in t the economy invests in the dirty technology if $k_t < \underline{k}^{TP}$.

In summary, the model reveals that different characteristics, as for example the existence of high transition costs or differences in the learning and capital intensities may lead to different tipping points for different economies. There is ample empirical evidence suggesting that it is worthwhile for an economy in early stages of development to use the most productive, but also the dirtiest technology to build-up physical capital. Once the economy is rich enough, it begins to adopt greener technologies, which are typically costly to implement, and carbon emissions eventually fall.²⁰ Applied to our model framework, this corresponds to the shift in technology from dirty to clean. Therefore, we will focus on the first tipping point and discuss the second one only where it is necessary.

2.6 Equilibrium dynamics

We define an equilibrium for the market economy as a sequence of prices $\{w_{i,t}, p_{i,t}\}_{t=0}^{\infty}$ and allocations $\{y_{i,t}, k_{i,t}, c_t^1, c_t^2\}_{t=0}^{\infty}$ with $i \in \{c, d\}$ such that the market for the final good clears, i.e.,

$$y_t = c_t^1 + c_t^2 + s_t,$$

the markets for factor inputs clear, which implies that all labor is hired, i.e. L = 1, and that investment in capital is equal to savings which leads to the following dynamic equation for capital:

$$k_{i,t+1} = s_t, \quad i \in \{c, d\}.$$
 (13)

To study the dynamics of the economy, we assume that $\gamma_c = 1$ and $\gamma_d \in (0,1)$, so we have $\gamma_c > \gamma_d$ as specified before.²¹ The assumption of $\gamma_c = 1$ implies that the

²⁰See e.g. Varvarigos (2014) and the references therein. Following Charlier et al. (2022), the technology shift from dirty to clean can also be interpreted as a change in the nature of the production process, which becomes less energy intensive as the economy develops; e.g., the transition from an economy based predominantly on manufacturing to one in which services dominate. The proposition that the transition to more environmentally friendly production methods occurs at higher stages of development is also consistent with the EKC hypothesis (Grossman & Krueger, 1995). Using an endogenous growth model with different technologies which vary in terms of in their dirtyness, Stokey (1998) provides an early theoretical explanation for the observed hump shaped pattern often found in empirical research. In our study, the transition in the pollution pattern is modelled as a technology shift from dirty to clean.

²¹The AK structure in clean production simplifies the growth dynamics and enhances the traceability of the model. However, our main results do not depend on the AK assumption, we could also assume a concave function for the clean technology where $\gamma_c > \gamma_d$ with $\gamma_c \in (0, 1)$ and $\gamma_d \in (0, 1)$.

marginal product of capital is constant and does not depend on the capital stock, reflecting the main property of a standard AK-model of endogenous growth (Romer, 1986). As a consequence, the return to capital of the clean technology is constant, allowing the economy to grow in the long run as the incentives for capital accumulation persist over time. The return to capital of the dirty technology, however, is decreasing over time which prevents the economy from achieving steady economic growth. From (3), (4), (8), (10), (11) and (13), the dynamics of the decentralized economy can be fully characterized by the equilibrium dynamics of physical capital and pollution:

$$k_{t+1} = \begin{cases} \frac{\beta}{1+\beta} \left[(1-\alpha_d) A_d k_t^{\gamma_d} + (1-\delta - \eta \varphi) k_t \right] & \text{if } k_t < \bar{k}^{TP}, \\ \frac{\beta}{1+\beta} \left[(1-\alpha_c) A_c (1-\theta) k_t + (1-\theta - \delta - \eta \varphi (1-\iota)) k_t \right] & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t = t^{TP}, \\ \frac{\beta}{1+\beta} \left[(1-\alpha_c) A_c k_t + (1-\delta - \eta \varphi (1-\iota)) k_t \right] & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t > t^{TP}, \end{cases}$$

$$\begin{cases} \varphi k_t & \text{if } k_t < \bar{k}^{TP}, \end{cases}$$

$$\Delta P_t = \begin{cases} \varphi(t - \iota)(1 - \theta)k_t & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t = t^{TP}, \\ \varphi(1 - \iota)k_t & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t > t^{TP}, \end{cases}$$

for given $P_0 > 0$, $k_0 > 0$ and with t^{TP} corresponding to the point in time at which the technology transition from dirty to clean occurs.

We now turn to the steady states of the economy under different technology choices. In order to show how the productivity of the clean technology determines the dynamic properties of the model, we will introduce two conditions to divide the values for A_c into three regions; each of which exhibits a different pattern of growth. Let us begin with the steady state under the dirty technology. From the dynamic equation for k_t , we find

$$k_d^{SS} = \left(\frac{(1-\alpha_d)A_d}{\beta^{-1}+\delta+\eta\varphi}\right)^{\frac{1}{1-\gamma_d}}.$$
(14)

In this steady state, all capital is allocated to the dirty technology and, due to diminishing returns to capital, long-term growth necessarily comes to an end. However, such a state only arises if the economy does not switch from the dirty to the clean technology along its transition path, which occurs if $\bar{k}^{TP} < k_d^{SS}$ or, equivalently,

$$\left(\frac{\alpha_d A_d}{\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}} < \left(\frac{(1-\alpha_d)A_d}{\beta^{-1}+\delta+\eta\varphi}\right)^{\frac{1}{1-\gamma_d}}.$$
(15)

This inequality represents the key condition for the transition from dirty to clean technologies to occur. We summarize this finding in Proposition 2.

Proposition 2. The economy can transition to the clean technology if the tipping point arises before the polluting steady state i.e. if $\bar{k}^{TP} < k_d^{SS}$.

We find that an economy is more likely to switch to clean production along its transition path, the lower the transition costs (θ), the lower the pollution intensity of capital (φ), the lower the sensitivity of the capital stock to climate change (η), the higher the discount factor (β) and the higher clean factor productivity (A_c). Beside that, a high value of α_c and low values of δ and α_d also promote a switch to clean production by increasing the likelihood that (15) is fulfilled.

The reasoning behind the impact of η goes as follows: Economies that are frequently affected by climate-induced capital losses cannot build up the minimal capital stock needed for technology transition. This is because the old generation, i.e., the investors, do not consider climate-related externalities in their investment decisions, which implies that the tipping points in the market economy does not depend on the parameters related to climate change while the dirty steady state does. Note that the determinants of the tipping points change as we turn to the social optimum of the economy.

Next, we express this condition by the TFP parameter of the clean technology. In doing so, we solve Inequality (15) for A_c :

$$A_c > \frac{\alpha_d(\beta^{-1} + \delta + \eta\varphi)}{(1 - \alpha_d)\alpha_c(1 - \theta)} \equiv \bar{A}_c^T,$$
(16)

where \bar{A}_c^T denotes a threshold value for productivity with the clean technology above which the tipping point arises before the steady state under the dirty technology. In other words, a productivity value higher than this threshold ensures that the clean technology can be adopted before the economy converges to the steady state with the dirty technology and we will refer to this condition as the "transition condition".

Next, we turn to the steady state of the clean technology. The production function of the clean technology is linear in capital and thus allows for sustained growth under the following condition:

$$\frac{k_{c,t+1}}{k_{c,t}} = \frac{\beta}{1+\beta} \left[(1-\alpha_c) A_c + (1-\delta - \eta \varphi (1-\iota)) \right] > 1,$$

which is, among others, more likely to hold the higher the technological upgrading in terms of pollution savings (ι). From that, we can derive a second condition for A_c , which reads

$$A_c > \left(\frac{1}{\beta} + \delta + \eta\varphi(1-\iota)\right) \frac{1}{1-\alpha_c} \equiv \bar{A}_c^G,\tag{17}$$

where \bar{A}_c^G corresponds to a second threshold for clean factor productivity above which the economy can sustain economic growth by investing in the clean technology. We call this condition the "green growth condition" in the following.



Figure 1: Clean factor productivity: threshold values and the corresponding regions

Having found two threshold values for clean factor productivity \bar{A}_c^T and \bar{A}_c^G , we show in the Appendix that under fairly general conditions it holds that $\bar{A}_c^G < \bar{A}_c^T$. This means that a relatively low clean factor productivity is sufficient for an economy to sustain economic growth with the clean technology, but a relatively high value for A_c is required to reach the tipping point.

We display the two thresholds in Figure 1, taking into account their order derived in the Appendix. The figure shows that the choice of technology and thus the dynamic properties of the economy depend on the value of A_c . In particular, we can distinguish three different regions for A_c . In Region 1, the clean technology is very productive and exceeds both thresholds. As a result, the transition condition is satisfied and the tipping point of the economy lies before the steady state under the dirty technology ($\bar{k}^{TP} < k_d^{SS}$), which ensures that the clean technology can be adopted before the economy fulfills the green growth condition and thus experiences sustained growth with the clean technology.

We provide the dynamics when A_c exceeds both thresholds in the phase diagram of Case 1 in Figure 2, where the savings function under the dirty technology is given by the concave function while the one under the clean technology by the linear function. For readability, we draw the second tipping point, \underline{k}^{TP} , only when it directly affects the growth dynamics, i.e., in Case 2 of Figure 2 and in Figure 3 further below. In Case 1, we see that regardless of the initial level of k, the economy accumulates sufficient capital to reach the tipping point, switches to clean production and experiences sustained growth with the clean technology.

Contrary, in Region 2 of Figure 1, the productivity parameter A_c is below both thresholds, implying that the clean technology is comparatively unproductive. Thus, such an economy does not meet either of the two conditions and converges to a unique steady state under the dirty technology, as is shown under Case 2 in Figure 2. Since the transition condition is not met, the economy's tipping point lies to the right of the steady state under the dirty technology ($\bar{k}^{TP} > k_d^{SS}$), which prevents the economy from making a successful transition to clean production. Moreover, due to its low factor productivity, the clean technology no longer implies sustained growth so that the capital formation



Case 1: Sustained growth with clean technology Case 2: Steady state with dirty technology

Figure 2: Phase diagrams

 \bar{k}^{TP}

 $\bar{k}^{TP} > k_d^{SS}$

 k_t

 k_d^{SS}

line of the clean technology now lies below the 45 degree line. As a result of these effects, the transitional dynamics of the economy now depend on its initial endowment with capital: Economies who are poorly endowed, i.e., those with less capital than \bar{k}^{TP} , will converge to the dirty steady state k_d^{SS} , only employing the dirty technology along their convergence path. Richer economies whose endowment exceeds the tipping point \bar{k}^{TP} will utilize the clean technology, but only for a limited time, since the clean technology is so unproductive that the capital stock declines over time until it falls below the tipping point \underline{k}^{TP} , whereupon a switch to the dirty technology occurs. From there on, the economy

again converges to the dirty steady state.

In Region 3 of Figure 1, the TFP value of the clean technology takes intermediate values, that is $\bar{A}_c^G < A_c < \bar{A}_c^T$, and the resulting outcome is a carbon trap. Economies with a productivity parameter A_c in between the two thresholds fulfill the green growth condition but not the transition condition. Although the clean technology is sufficiently productive to generate permanent economic growth, the violation of the transition condition implies that the steady state with dirty technology arises prior to the tipping point $(\bar{k}^{TP} > k_d^{SS})$, which prevents the economy from conducting a technology switch from dirty to clean.

The phase diagram for such values of A_c can be found in Case 3 of Figure 2 where the initial capital endowment again decides about the long-term prospects of the economy. Similar to Case 2, an initial value of capital below \bar{k}^{TP} will set the economy on a convergence path to the dirty, low-income steady state, k_d^{SS} , although green growth would be technologically feasible. We refer to such a constellation as a "carbon trap", where the trap region is given by the interval between the origin and \bar{k}^{TP} . Moreover, we refer to the interval between k_d^{SS} and \bar{k}^{TP} as the "upper trap region" (indicated by the ocher dashed arrow in Figure 2), which denotes the critical interval that an economy must overcome for the technology transition from dirty to clean to occur, and we examine how policies can affect the size of this upper trap region in Section 4. In line with our previous findings, the size of the upper trap region is larger for economies with a high climate exposure since $\partial k_d^{SS}/\partial \eta < 0$ while $\partial \bar{k}^{TP}/\partial \eta = 0$ in the market economy. For wealthier economies the outcome is much more optimistic, as an endowment higher than the tipping point propels them on a sustained growth path with the clean technology.

We summarize our findings in the following proposition.

Proposition 3. Depending on the parameter value for A_c , we can distinguish three possible patterns of growth:

Case 1: The economy is subject to sustained growth with the clean technology if its productivity A_c is (sufficiently) high such that $A_c > \bar{A}_c^T > \bar{A}_c^G$.

Case 2: The economy converges to a unique dirty steady state if its clean factor productivity A_c is (sufficiently) low such that $A_c < \bar{A}_c^G < \bar{A}_c^T$.

Case 3: If clean factor productivity A_c is subject to intermediate values such that $\bar{A}_c^G < A_c < \bar{A}_c^T$, a carbon trap exists and the economy will converge to this trap when the initial level of capital k_0 is lower than \bar{k}^{TP} .

Let us consider the opposite threshold order, that is, $\bar{A}_c^G > \bar{A}_c^T$ and the resulting outcomes for the economy. Then, the types of possible steady states change: The carbon trap disappears and instead permanent cyclical fluctuations emerges, while sustained growth with the clean technology and the steady state with the dirty technology remain



Figure 3: Phase diagram with cyclical fluctuations

as possible growth patterns for the economy. If $A_c > \bar{A}_c^G > \bar{A}_c^T$ holds, then the economy will experience continuous growth under the clean technology, as in Case 1 of Figure 2. However, a novel growth pattern arises if A_c exceeds \bar{A}_c^T but not \bar{A}_c^G , that is, if the transition condition is satisfied but not the green growth condition. Thus, the tipping point of the economy is to the left of the steady state under the dirty technology ($\bar{k}^{TP} < k_d^{SS}$), allowing the economy to switch to the clean technology along its transition path. However, as the green growth condition is not met, the capital stock shrinks over time as the productivity of the clean technology is too low to enable long-term growth. Due to this contraction, the economy will eventually pass below \underline{k}^{TP} and return to the dirty technology sooner or later. This process is repeated and permanent cyclical fluctuations emerge, as illustrated in Figure 3. Once A_c is below both threshold values, the steady state under the dirty technology arises and the relative size of \bar{A}_c^G and \bar{A}_c^T becomes irrelevant, which corresponds to Case 2 in Figure 2.

We summarize our findings in the following proposition:

Proposition 4. The set of possible long-run dynamics for the economy depends on the relative size of \bar{A}_c^T and \bar{A}_c^G :

- i. If $\bar{A}_c^T > \bar{A}_c^G$ holds, then the economy inhibits the growth dynamics described in Proposition 3.
- ii. If $\bar{A}_c^T < \bar{A}_c^G$ holds, then the economy inhibits the growth dynamics described in Propo-

sition 3, whereby permanent cyclical fluctuations emerge instead of a carbon trap.

3 Social planner

Next, we turn to the social optimum for the economy. Given the model framework, nonmarket effects i.e., negative and positive externalities are a crucial part of the analysis. The social planner takes into account the positive learning effects in capital accumulation from both the clean and dirty technology as well as the negative externality related to climate change. In addition, the social planner internalizes that part of the capital stock becomes obsolete when the economy transitions from the dirty to the clean technology, which reduces the stock of capital available for future generations. Accordingly, the maximization problem of the social planner reads

$$\begin{split} & \max_{\{c_t^1, c_{t+1}^2, k_{t+1}, k_{d,t}, k_{c,t}\}_{t=0}^{\infty}} W = \sum_{t=0}^{\infty} \beta_s^t \left(\log(c_t^1) + \beta \log(c_{t+1}^2) \right) \\ & \text{subject to} \\ & c_t^1 + c_t^2 + k_{t+1} - (1 - \delta - \eta \varphi) k_{d,t} - \left(1 - \frac{\delta + \eta \varphi(1 - \iota)}{(1 - \theta)^{\kappa}} \right) k_{c,t} = A_c k_{c,t}^{\gamma_c} + A_d k_{d,t}^{\gamma_d} \\ & k_t = \frac{k_{c,t}}{(1 - \theta)^{\kappa}} + k_{d,t}, \\ & k_0 > 0, \end{split}$$

where $\beta_s \in (0, 1)$ is the social planner's discount factor and W denotes social welfare. Furthermore, κ accounts for the transition costs and takes the value of either zero or one, depending on whether there is a transition from the dirty to clean technology or vice versa. In each period, the social planner can decide on the capital stock in the next period k_{t+1} , which can be allocated to either the clean or the dirty technology. Accordingly, we maximize for the aggregate capital stock over k_{t+1} while for the technology-specific stocks over $k_{c,t}$ and $k_{d,t}$ and we provide a detailed solution to the social planner's maximization problem in the Appendix. Any solution must also satisfy the transversality condition. By setting up the Lagrangian and combining the first-order conditions, it follows that

$$(1-\theta)^{\kappa} \left(\gamma_c A_c k_{c,t+1}^{\gamma_c - 1} + 1 - \frac{\delta - \varphi \eta + \iota \eta \varphi}{(1-\theta)^{\kappa}} \right) = \gamma_d A_d k_{d,t+1}^{\gamma_d - 1} + 1 - \delta - \eta \varphi.$$
(18)

To draw comparisons to the market outcome, we again assume that $\gamma_c = 1$ whereas $\gamma_d \in (0, 1)$, so that the previous equation changes to

$$(1-\theta)^{\kappa}(1+A_c) - \delta - \eta\varphi(1-\iota) = \gamma_d A_d k_{d,t+1}^{\gamma_d-1} + 1 - \delta - \eta\varphi.$$
⁽¹⁹⁾

Let us study the tipping point for the transition from dirty to clean, so that $\kappa = 1$. The social returns to both technologies are the same, and the social planner is indifferent between investing the entire capital stock in the clean or the dirty technology if:

$$\bar{k}^{SP} = \left(\frac{\gamma_d A_d}{(1-\theta)A_c - \theta + \eta\varphi\iota}\right)^{\frac{1}{1-\gamma_d}}.$$
(20)

Since the right-hand side of (19) is falling in k, this tipping point is the lower bound for the technology transition from dirty to clean to occur.

Let us discuss the factors that determine this tipping point in turn. The term $\gamma_d A_d$ reflects the internalized learning effects of the dirty technology that increase the stock of capital after which the social planner decides to invest in the clean technology. The reverse holds for the term $(1 - \theta)A_c$ that represents the internalized learning spillovers of the clean technology. Since a change to the clean technology is associated with costs, however, these spillovers are reduced by the factor $(1 - \theta)$. The term $-\theta$ also increases the tipping point, as it captures the fact that changing to the clean technology today makes some of today's aggregate capital stock obsolete and thus will reduce the capital that is available tomorrow. This effect is not internalized by agents in the market economy, because they do not care what happens to the capital stock after they are old. Finally, unlike the market economy, the social planner takes into account the effects related to climate change.

In particular, the higher the pollution intensity of capital (φ) and the greater the sensitivity of the capital stock to climate change (η) , the lower is the tipping point of the social planner. Moreover, the higher the technological enhancement in terms of pollution savings (ι) , the more worthwhile a technology switch and thus the lower the socially optimal tipping point. Using standard parameter values, we numerically demonstrate in Section 5.2 that the tipping point of the social planner is lower that the market tipping point, i.e. $\bar{k}^{SP} < \bar{k}^{TP}$.

Similar to the decentralized economy, we can derive a second tipping point for the transition from clean to dirty by setting $\kappa = 0$ in Equation (19). Thus, we obtain

$$\underline{k}^{SP} = \left(\frac{\gamma_d A_d}{A_c + \eta \varphi \iota}\right)^{\frac{1}{1 - \gamma_d}}$$

If the social planner has invested in the clean technology and the aggregate capital stocks passes below \underline{k}^{SP} , a transition from the clean to the dirty technology will occur. Note that this tipping point is unambiguously smaller than the corresponding tipping point in the market economy, i.e., $\underline{k}^{SP} < \underline{k}^{TP}$.

4 Policy analysis

We consider and compare two policies to implement the social planner's tipping point in the market economy and ask whether such policies can prevent economies from falling into a carbon trap. In addition, we assess whether these policies can reduce the upper trap region and thus pave the way for trapped economies to achieve sustained green growth. The first policy is a carbon tax imposed on dirty production and whose proceeds are given to the young. The second instrument is a subsidy granted to the clean technology and financed by an income tax on the young generation. Whether we make the green technology more productive or the dirty technology less productive affects capital accumulation differently and yields new insights regarding the dynamic properties of the model. Both policies are aimed at providing an incentive for households to invest in the clean technology in an optimal way with respect to the development level of the economy. However, these policies do not guarantee that the market economy will follow the same transition path as in the social optimum. They merely ensure that the market economy switches to the clean technology for a level of capital that corresponds to the one chosen by the social planner. In what follows, we refer to such a policy as being an "optimal instrument" even though the transition dynamics are not fully considered.

4.1 Carbon tax

Firms must pay a carbon tax proportional to the pollution they generate. Since pollution is complementary to output, the carbon tax is equivalent to a tax on dirty production. Let τ_d stand for a constant tax rate imposed by the government on dirty production. Under this tax, output with the dirty technology is given by

$$y_{d,t} = (1 - \tau_d) A_d k_{d,t}^{\gamma d}$$

When maximizing profits, firms incorporate the tax in their decision making so that the factor prices change accordingly. The modified prices for labor and capital are summarized in the appendix. Assuming again that the household chooses the investment opportunity whose return is higher than the alternative, we obtain the following tax-adjusted tipping point:

$$\tilde{k}^{TP} = \left(\frac{(1-\tau_d)\alpha_d A_d}{\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}}.$$
(21)

From that, we can observe that \tilde{k}^{TP} and τ_d are inversely related and thus $\tilde{k}^{TP} < \bar{k}^{TP}$. Therefore, a higher carbon tax lowers the critical capital stock and thus promotes a switch in favor of the clean technology. Along with imposing a tax on dirty production, the government redistributes the tax revenue to the contemporary young agents as a lump-sum transfer so that the government's budget will be balanced. Accordingly, when operating with the dirty technology, the intertemporal budget constraint and the optimal saving function change to

$$w_{d,t} + \tau_d y_{d,t} + (1 - \theta \psi - \Lambda_d) k_t = c_t^1 + \frac{c_{t+1}^2}{r_{d,t+1}}, \text{ and} \\ s_t = \frac{\beta}{1 + \beta} \left[w_{d,t} + \tau_d y_{d,t} + (1 - \theta \psi - \delta - \eta \varphi) k_t \right].$$

The tax has a positive impact on savings and thus fosters capital formation with the dirty technology. While the dynamic equation for pollution and capital formation in the case of $k_t \geq \bar{k}^{TP}$ remains the same as in the baseline model, capital accumulation under the policy changes as follows:

$$k_{t+1} = \frac{\beta}{1+\beta} \left[(1 - \alpha_d + \alpha_d \tau_d) A_d k_t^{\gamma_d} + (1 - \delta - \eta \varphi) k_t \right] \quad \text{if } k_t < \bar{k}^{TP},$$

for given $k_0 > 0$. Overall, the introduced carbon tax promotes the shift from dirty to clean technology twofold: First, by reducing the tipping point and second by fostering capital formation so that the economy is more likely to be able to build the capital stock needed to reach the (reduced) tipping point. These two effects have an impact on both the level of A_c , which enables the transition to the clean technology, as well as on the upper trap region, which is shown below.

With regard to the threshold value \bar{A}_c^T , we find that the carbon tax appears in two instances in the relevant expression

$$A_c > \frac{(1 - \tau_d)\alpha_d(\beta^{-1} + \delta + \eta\varphi)}{(1 - \alpha_d + \alpha_d\tau_d)\alpha_c(1 - \theta)} \equiv \tilde{A}_c^T,$$
(22)

where it reduces the threshold from \bar{A}_c^T to \tilde{A}_c^T .²² Therefore, the transition condition is satisfied for a wider range of values for A_c , making it more likely for less productive economies to switch to the clean technology along their transition paths. The green growth condition described by \bar{A}_c^G remains unchanged compared to the model without policy intervention.



Figure 4: Threshold values and regions with a carbon tax

Figure 4 indicates the new threshold associated with the modified transition condition.

 $^{^{22}}$ The tax-adjusted steady state of the dirty technology is given by Equation (34) in the Appendix.

The carbon tax makes Region 1 (where long-term growth with the clean technology exists) broader and Region 3 (carbon traps) narrower. Thus, a carbon tax whose proceeds are redistributed to the young generation helps a capital-poor economy, where green growth would be technologically feasible, to transition to clean production and thus to escape the carbon trap.

In the following, we show that a carbon tax constrains not only the set of values for A_c under which a carbon trap exists but also the interval for capital endowments that pitch the economy into the carbon trap. The reason for this is twofold: First, the tax policy gives households an incentive to adopt the clean technology by lowering the tipping point from \bar{k}^{TP} to \tilde{k}^{TP} , and second, it provides workers with an incentive to save because they receive the redistributed tax revenues that shift the capital accumulation equation upward from $k_{d,t+1}$ to $\tilde{k}_{d,t+1}$, raising the dirty steady state from k_d^{SS} to \tilde{k}_d^{SS} . We illustrate both effects in Figure 5, where upper trap region shrinks from $[k_d^{SS}, \bar{k}^{TP}]$ to $[\tilde{k}_d^{SS}, \tilde{k}^{TP}]$ and consequently the set of capital values from which the economy converges to the carbon trap becomes smaller.

We summarize our findings in the following proposition:

Proposition 5. A tax on the dirty technology whose proceeds are transferred to the young has the following effects:

- 1. It reduces the set of values for A_c under which a carbon trap emerges in the economy.
- 2. Given that a carbon trap exists, the tax narrows the upper trap region from $[k_d^{SS}, \bar{k}^{TP}]$ to $[\tilde{k}_d^{SS}, \tilde{k}^{TP}]$.

To find the optimal tax, we equalize the tax-adjusted market tipping point derived in (21) with the tipping point of the social planner derived in (20) such that $\tilde{k}^{TP} = \bar{k}^{SP}$. Solving yields

$$\tau_d^* = 1 - \frac{\gamma_d \alpha_c (1 - \theta) A_c}{\alpha_d \left((1 - \theta) A_c - \theta + \eta \varphi \iota \right)}, \quad \in (0, 1).$$
(23)

In light of Equation (23), we can conclude that the optimal tax depends, among others, positively on the parameters related to climate change: The higher the technological upgrading in terms of pollution savings (ι) , the higher the pollution intensity of capital (φ) , and the greater the sensitivity of capital stock to climate change (η) , the higher the optimal carbon tax that induces households to adopt the clean technology at the socially optimal level of development.



Figure 5: Phase diagram: Carbon trap with tax policy

4.2 Subsidy

As a second policy instrument we consider a subsidy strategy. Let τ_c denote a timeinvariant subsidy granted to the clean technology such that

$$y_{c,t} = (1 + \tau_c) A_c k_{c,t}.$$

Under the same assumptions as for the tax and with an analogous procedure, we obtain the following tipping point:

$$\hat{k}^{TP} = \left(\frac{\alpha_d A_d}{(1+\tau_c)\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}}$$

Intuitively, a subsidy on clean production lowers the tipping point and thus $\hat{k}^{TP} < \bar{k}^{TP}$. The subsidy is financed by an income tax on the young such that the government runs a balanced budget in every period. Given that the clean technology is employed, the intertemporal budget constraint and the optimal saving function of the household change

$$w_{c,t} - \tau_c y_{c,t} + (1 - \theta \psi - \Lambda_c) k_t = c_t^1 + \frac{c_{t+1}^2}{r_{c,t+1}}, \text{ and}$$
$$s_t = \frac{\beta}{1+\beta} [w_{c,t} - \tau_c y_{c,t} + (1 - \theta \psi - \delta - \eta \varphi (1 - \iota)) k_t]$$

Given the policy design, a subsidy impacts both the tipping point and capital accumulation. While the dynamic equations for the change in pollution and capital accumulation with the dirty technology remain the same as in the baseline framework, capital with the clean technology evolves over time according to

$$k_{t+1} = \begin{cases} \frac{\beta}{1+\beta} \left[(1 - \alpha_c - \alpha_c \tau_c) A_c (1 - \theta) k_t + (1 - \theta - \delta - \eta \varphi (1 - \iota)) k_t \right] & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t = t^{TP}, \\ \frac{\beta}{1+\beta} \left[(1 - \alpha_c - \alpha_c \tau_c) A_c k_t + (1 - \delta - \eta \varphi (1 - \iota)) k_t \right] & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t > t^{TP} \end{cases}$$

for given k_0 . Subsidizing clean production has an ambiguous effect on the tipping point and capital formation: A higher level of subsidy τ_c facilitates the technology transition by reducing the tipping point from \bar{k}^{TP} to \hat{k}^{TP} . On the other hand, a too high subsidy in a relatively poor country dampens capital accumulation with the clean technology and thus future growth prospects preventing an economy to conduct a technology switch from dirty to clean. The first of those effects is captured by the transition condition:

$$A_c > \frac{\alpha_d(\beta^{-1} + \delta + \eta\varphi)}{(1 - \alpha_d)(1 + \tau_c)\alpha_c(1 - \theta)} \equiv \hat{A}_c^T,$$
(24)

where the subsidy reduces the threshold from \bar{A}_c^T to \hat{A}_c^T . All else remaining the same, it is straightforward to see that this inequality is more likely to hold in the presence of a subsidy. Thus, the implementation of such a policy increases the likelihood that the economy can pass the critical capital stock and not get stuck with the dirty technology in an early stage of development. However, unlike the tax policy, the subsidy affects the green growth condition

$$A_c > \left(\frac{1}{\beta} + \delta + \eta\varphi(1-\iota)\right) \frac{1}{(1-\alpha_c - \alpha_c\tau_c)} \equiv \hat{A}_c^G,$$
(25)

by increasing the threshold from \bar{A}_c^G to \hat{A}_c^G and thus constraining the set of A_c values under which this condition is fulfilled. Hence, a subsidy policy designed in this way, has a negative impact on a country's economic development. This is because the public funds needed to finance the subsidy are paid by young people, reducing their disposable income and thus their savings, which has a negative impact on future capital formation and growth. The subsidy narrows Region 3 (carbon traps) but extends Region 2 (nongrowing dirty steady state). Figure 6 visualizes the countervailing effects of the subsidy

to

policy by means of the modified threshold values.



Figure 6: Threshold values and regions with subsidy

While the subsidy should be high enough to incentivize the adoption of the clean technology, overly stringent policies may deprive a country of its economic ability to escape the carbon trap and may encourage the use of the dirty technology in the long-run.

As in the case of the carbon tax, the effect of the subsidy is not limited to the values of A_c that lead to a carbon trap. The subsidy also influences the set of initial capital endowments under which an economy converges to the carbon trap, as we show in Figure 7. Due to the policy's impact on the transition condition, the upper trap region shrinks from $[k_d^{SS}, \bar{k}^{TP}]$ to $[k_d^{SS}, \hat{k}^{TP}]$. Note, however, that this reduction of the upper trap region is smaller than under the tax policy, since there is no additional capital formation under the dirty technology that helps the economy to overcome the upper trap region. In addition, the negative impact of the subsidy on capital formation stemming from the green growth condition translates into a decline of the capital accumulation equation under the clean technology from $k_{c,t+1}$ to $k_{c,t+1}$. If this capital dampening effect were so pronounced that the capital accumulation function fell below the 45 degree line, the economy would only have a steady state with the dirty technology, corresponding to the growth pattern of Case 2 in Figure 2. In such a case, the capital stock shrinks under the clean technology and, regardless of the initial capital stock, the economy converges to k_d^{SS} . Clearly, such a subsidy would not be optimal but it demonstrates that climate policy should be carefully calibrated, keeping in mind that the funds used to promote clean technologies need to be financed, which reduces the resources available for capital investments and thus adversely affects future growth prospects.

We summarize our findings related the subsidy policy in Proposition 6.

Proposition 6. Providing subsidies to the clean technology (by financing the subsidies through an income tax on the young generation) has the following effects:

- 1. It reduces the set of A_c -values under which a carbon trap exists in the economy.
- 2. Given that a carbon trap exists, it reduces the upper trap region from $[k_d^{SS}, \bar{k}^{TP}]$ to $[k_d^{SS}, \hat{k}^{TP}]$. However, this reduction is smaller than under the tax policy as k_d^{SS} remains unaffected.



Figure 7: Phase diagram: Carbon trap with subsidy

The optimal subsidy τ_c^* is obtained by comparing the socially optimal tipping point with the tipping point that arises in the market economy under the subsidy policy, i.e. $\hat{k}^{TP} = \bar{k}^{SP}$, which yields

$$\tau_c^* = \frac{\alpha_d \left((1-\theta) A_c - \theta + \eta \varphi \iota \right)}{\gamma_d \alpha_c A_c (1-\theta)} - 1, \quad \in (0,1).$$
(26)

The optimal subsidy is, among other things, an increasing function of the parameters related to climate change: a high pollution intensity (φ), a high capital sensitivity (η) and high pollution savings (ι) generate a higher subsidy in order to foster adoption in the market economy.

In the policy analysis so far, we have assumed that the young household receives the revenues from the tax on the dirty technology and finances the subsidies for the clean technology, leading to the following key conclusion: For an economy locked in a carbon trap our analysis reveals that the tax policy is better suited to trigger the socially optimal tipping point because unlike the subsidy, the public funds released by the carbon tax provide an additional stimulus to capital accumulation which further reduces the upper trap region.

Instead, one could assume that the old generation receives the revenues from the tax and has to pay for the subsidy. We do so in the Appendix and find that these policies lead to a "wage effect" and an "anticipation effect," both of which affect the incentives to accumulate capital. Given that tax revenues are transferred to the old generation, the "wage effect" states that a firm can pay lower wages to young households because of the tax burden, which reduces their disposable income and thus their incentive to save and accumulate capital. Moreover, young households anticipate the increase in disposable income in old age so that their incentives to save for old-age consumption is reduced, which in turn lowers capital accumulation. This corresponds to the "anticipation effect".

Overall, by shifting resources from the young to the old, such a tax policy not only reduces the tipping point but also the incentives for capital accumulation, reflecting the widespread concern that (overly) stringent climate policies could harm economic development and thus possibly delay rather than accelerate the technology transition.²³ Using standard parameter values, we show in the Appendix that the negative effect of the tax on capital accumulation quantitatively outweighs the tax-induced reduction in the tipping point, enlarging the overall size of the upper trap region. This illustrates that the implementation of well-intended climate policies in capital-poor economies should take into account potential unintended side effects that may weaken incentives for capital accumulation and thus favour the emergence of carbon traps.

When the subsidy is financed by the old generation, the same effects occur as with the tax policy, where both the wage– and the anticipation effect stimulate capital accumulation and, together with the reduced tipping point, encourage the adoption of clean technologies. More specifically, the young anticipate that they will have to finance the clean technology in old age and compensate for this decrease in disposable income by increasing capital formation, which represents the "anticipation effect." Together with the subsidy-induced increase in wages, i.e., the "wage effect", capital formation is stimulated, which promotes the adoption of the clean technology as a whole.

We summarize our results for all constellations of policies and redistribution schemes:

- 1. Regardless of the policy and redistribution scheme, the tipping point \bar{k}^{TP} is reduced.
- 2. While every policy that removes resources from the young, reduces capital accumulation, the opposite holds for policies that take resources from the old.
- 3. The tax on the dirty technology leaves A_c^G unaffected while the subsidy for the clean technology leaves k_d^{SS} unaffected.

 $^{^{23}}$ See e.g. Charlier et al. (2022) for a more detailed discussion.

5 Numerical simulation

Above, we have analytically demonstrated the difference between the tipping points in the social optimum and the market economy. However, in order to determine by how many periods the tipping points are apart, we have to resort to a numerical simulation of the model.

5.1 Calibration

We rely on various data sources for the calibration of the parameter values. The length of one period is around 30 years. We set capital depreciation to $\delta = 0.3$ which corresponds to a depreciation rate of 4% per annum. The discount factor β is equal to 0.7; slightly higher than in Dao & Edenhofer (2018) who employed an OLG model to examine fiscal strategies to avoid poverty-environment traps. Since green technologies tend to be relatively capital intensive, we set $\alpha_c = 0.35$, while we assume a standard capital intensity of $\alpha_d = 0.3$ for the polluting technology (IEA, 2022). The learning intensities are $\nu_c = 0.65$ and $\nu_d = 0.4$, reflecting the gap in learning rates between clean and dirty technologies in the empirical literature, see e.g. Rubin et al. (2015). Concerning the pollution intensity of capital, we assume $\varphi = 1.11$ as in Dao & Edenhofer (2018). For the transition costs in terms of capital we rely on Pommeret & Schubert (2009) and assume $\theta = 0.15$. For climate change vulnerability, we set $\eta = 0.3$ in line with Bretschger & Komarov (2023).²⁴ With a technology switch from dirty to clean, the economy can reduce its pollution intensity by 90%, so that $\iota = 0.9.^{25}$ For the TFP values of the technologies, we impose that the clean technology lags relatively behind the dirty technology, assuming that A_c is between 2.7 and 4, while A_d is set to 5. The range of A_c values is used below to simulate the different growth patterns presented in the theoretical part. We summarize the parameter values in Table 2.

5.2 Simulation

Using the parameter values described above and setting $A_c = 4$ leads to unbounded green growth in the long run, as the productivity value of the clean technology exceeds the two productivity thresholds given by $\bar{A}_c^G = 2.7$ and $\bar{A}_c^T = 3$. Accordingly, the market economy can adopt the clean technology regardless of its initial capital endowment. The evolution of the capital stock and total consumption for the social planner and the market economy is shown in Figure 8. We find that the socially optimal tipping point occurs 3 periods, i.e., 90 years, before the decentralized tipping point, so that the social planner switches

²⁴For economies that are particularly exposed to the consequences of global warming, we set $\eta = 0.5$ and demonstrate in the appendix that this can lead to a climate-induced carbon trap.

 $^{^{25}}$ We also investigated the case of a pollution-free technology, i.e., $\iota=1$ but this does not change our results.

Parameter	Description	Value
β	Discount factor	0.7
$lpha_d$	Capital intensity dirty technology	0.3
$lpha_c$	Capital intensity clean technology	0.35
$ u_d$	Learning intensity dirty technology	0.4
$ u_c$	Learning intensity clean technology	0.65
δ	Depreciation of capital	0.3
arphi	Pollution intensity	1.11
η	Impact intensity	0.3
A_c	Productivity clean technology	2.7-4
A_d	Productivity dirty technology	5
heta	Transition costs	0.15
l	Pollution savings	0.9

Table 2: Parameter va	lues
-----------------------	------

to clean production significantly earlier than the market economy. There are two reasons for this result: First, the critical capital stock at which the social planner transitions to the clean technology ($\bar{k}^{SP} = 1$) is smaller than the market tipping point ($\bar{k}^{TP} = 2.2$), reflecting that the social planner internalizes all externalities (including the transition costs). Due to the non-internalized transition costs in the market economy, capital accumulation flattens out at the time of transition, i.e., in $t^{TP} = 6$, which temporarily delays economic development (Panel A).



Figure 8: Sustained growth with clean technology

Second, the social planner accumulates capital much faster than the market economy, both before and after the tipping point occurs. This fast capital accumulation and resulting rapid transition to clean production leads to a large increase in consumption in the long run, but at the expense of low consumption in the first 2 periods (≈ 60 years), during which consumption in the social optimum is lower than in the market economy (Panel B). To isolate the sole effect of the external effects on the transition timings, we modify the saving function of the market economy later in this section.

We provide further insights into the investment decisions of the social planner and the market economy in Figure 9. As illustrated in Panel A and Panel B, aggregate consumption and savings in the market economy are a constant fraction of total wealth, amounting to 0.59 and 0.41 respectively. In contrast, we find that the socially optimal savings rate is already significantly higher in the first period under the dirty technology, taking a value of 0.62. Moreover, this savings rate temporarily jumps to 0.68 in the period before the tipping point occurs and then adopts — from the tipping point period onward - a constant value of 0.65. We observe the opposite pattern for the socially optimal consumption rate (Panel B) and consumption growth rate (Panel C), respectively, with both dropping in the period before the technology transition occurs. The reason for these temporal peaks lies in the transversality condition of the social planner as well as in the tipping point condition: As is common in AK-models, the ratio of consumption to capital takes a fixed value that is determined by the aforementioned transversality condition. To achieve this desired ratio, the social planner needs to increase savings to compensate for the additional capital losses resulting from the transition to clean production. The growth rate of consumption in the market economy steadily decreases until the tipping point is reached (Panel C), reflecting the concave production structure of the dirty technology. After passing the tipping point, the growth rate of consumption is 0.35%, well below the growth rate of consumption in the social optimum, which is 2.26%. Regarding the optimal policies, our simulation reveals that a tax rate of $\tau_d^* = 0.21$ or a subsidy of $\tau_c^* = 0.27$ can implement the socially optimal tipping point in the market economy.



Figure 9: Sustained growth with clean technology

Next, we illustrate the case of a carbon trap. In doing so, we leave the other parameter values unchanged and lower the TFP value of the clean technology to $A_c = 2.75$ so that the green growth condition is satisfied ($\bar{A}_c^G = 2.7$), but not the transition condition ($\bar{A}_c^T = 3$).²⁶ The resulting plots are shown in Figure 10, where the tipping period in the market economy, t^{TP} , vanishes because the transition condition is not satisfied and thus the economy cannot transition to clean production in the market economy in the absence of climate policies. The upper trap region resulting from the simulation is shown in Panel C and is determined by the interval between $\bar{k}^{TP} = 7.5$ and $k_d^{SS} = 5.8$. All initial values of capital below $\bar{k}^{TP} = 7.5$ form the trap region from which the economy converges to the dirty steady state with a convergence time of about 20 periods in the present illustration.²⁷ The consumption growth rate shown in Panel D also follows this pattern and eventually converges to zero.²⁸ Initial values of capital above $\bar{k}^{TP} = 7.5$ allow the economy to generate sustained growth with the clean technology.



Figure 10: Carbon trap

The social optimum changes only slightly compared to the previous simulation with the higher TFP value. The social planner again switches to clean production after three periods, and experiences sustained green growth in the long run. In this process, the economy consumes less than the market economy until the technology switch occurs, i.e., for 3 periods (Panel B). The socially optimal growth rate of consumption after the

²⁶In the Appendix, we show that a carbon trap can also arise from a combination of higher climate exposure and lower clean factor productivity, where the corresponding parameter values are $\eta = 0.5$ and $A_c = 3$.

 $^{^{27}}$ The convergence time depends on the initial stock of capital and does not hold generally.

 $^{^{28}}$ We omit the case where the steady state with dirty technology is the only long-run outcome, as this is just a generalization of the case of a carbon trap, where convergence to the trap is global.
transition is about 1.39% and thus lower compared to the previous simulation where the higher A_c -value was employed (Panel D). By accumulating sufficient capital and internalizing all non-market effects, the social planner ensures, unlike the market economy, that the transition condition is satisfied.²⁹ However, the gap between $\bar{k}^{TP} = 7.5$ and $\bar{k}^{SP} = 3.1$ in Panel C can be closed using the optimal policies derived in Section 4. Specifically a tax of $\tau_d^* = 0.23$ or a subsidy of $\tau_c^* = 0.3$ can implement the socially optimal tipping point in the market economy and thus ensure that the economy can switch to clean production and not get stuck in the carbon trap in the long run.

Finally, we perform the simulations for the case of cyclical fluctuations. Increasing the capital intensity of the clean technology to $\alpha_c = 0.4$ leads to a reversal of the productivity thresholds, which are now given by $\bar{A}_c^T = 2.6$ and $\bar{A}_c^G = 2.95$. Specifying $A_c = 2.7$, gives rise to cyclical fluctuations, which is illustrated in Figure 11. As in the previous simulations, the social planner switches to clean production after three periods. As shown in Panel B, consumption in the social optimum is lower than in the market economy for about 90 years. In Panel C we see that the capital stock of the market economy oscillates between the two tipping points introduced in the theoretical section. Initially, the capital stock increases until the tipping point ($\bar{k}^{TP} = 5.1$) is reached after 10 periods, whereupon the economy switches to the clean technology for the first time. However, this transition leads to a contraction of the capital stock due to the low productivity with the clean technology until the economy falls below $k^{TP} = 2.9$ and returns to dirty production after 18 periods. Thereafter, total capital rises again until the economy shifts back to the green technology; this process repeats itself creating a stagnating economy with recurring technology cycles.³⁰ As shown in Panel B, total consumption follows the same pattern. Accordingly, the growth rate of consumption shown in Panel D fluctuates around zero, taking positive and negative values intermittently, while the growth rate of consumption in the social optimum is constant, adopting a value of 1.36%.

5.3 Simulation with modified saving function

As mentioned before, one of the two possible reasons why the social planner switches to clean production earlier than the market economy is its higher savings rate. To control for this channel and to determine the extent to which the difference in adoption times is due to the difference in actual tipping values and the involved externalities rather than savings rates, we modify our model by including an additional term in the utility function:

$$U_t(c_t^1, c_{t+1}^2, s_t) = \log(c_t^1) + \beta \log(c_{t+1}^2) + \epsilon \log(s_t).$$

²⁹In Panel C, we see that $\bar{k}^{SP} < k_d^{SS}$ while $\bar{k}^{TP} > k_d^{SS}$, with the corresponding values given by $\bar{k}^{TP} = 7.5$, $\bar{k}^{SP} = 3.1$ and $k_d^{SS} = 5.8$.

³⁰Note that the optimal policies derived in Section 4 cannot provide a way out of stagnation, since the obstacle to long-term green growth is not the transition to clean production, but the unproductive green technology.



Figure 11: Cyclical fluctuations

Hence, agents have an explicit preference for savings now, reflecting some "warm-glow" or additional altruism motive; the strength of this effect is captured by the parameter ϵ .³¹ Given this modification, savings are now given by

$$s_t = \frac{\beta + \epsilon}{1 + \beta + \epsilon} \left[w_{i,t} + (1 - \theta \psi - \Lambda_i) k_t \right], \quad i \in \{c, d\}.$$

In the following, the obtained results with this modified savings function are compared to the case of sustained growth with the clean technology, in which the original savings function was applied (see Figure 7 and 9). In doing so, we use the parameter values listed in Table 2 and again set $A_c = 4$. Thus, growth occurs in both the market economy and the social optimum in the long run, regardless of the initial capital endowment of the economy. The modified savings function enhances capital accumulation in the market economy while the tipping points given by $\bar{k}^{SP} = 1$ and $\bar{k}^{TP} = 2.2$ as well as the social planner solution remain unchanged.³² The parameter ϵ is calibrated to align the savings rate of the market economy with that of the social optimum, which yields $\epsilon = 1.15$.

We see in Figure 12 that the savings rate (0.65) and consumption rate (0.35) of the social planner and the market economy displayed in Panel D and Panel E coincide, ensuring that any differences in adoption times are now entirely attributable to the nonmarket effects and the resulting differences in the tipping capital stocks. The temporary

 $^{^{31}}$ We assume that the social planner does not internalize this altruism, as otherwise the relative difference in savings rates between the social optimum and the market economy would remain the same.

³²As a result, the optimal tax ($\tau_d^* = 0.21$) and subsidy ($\tau_c^* = 0.27$) to implement the social planner's tipping point in the market economy remain also unchanged.



Figure 12: Simulation with modified savings function

increase (decrease) in the savings rate (consumption rate) of the social planner in the second period arises again due to the compensation of the transition costs. Figure 12 shows that despite the same propensity to save as in the social optimum, the market economy transitions to the clean technology later than socially optimal, with the time difference now being reduced to a single period (≈ 30 years). This suggests that differences in savings behavior is not the only factor delaying the adoption of green technologies in the market economy; the non-internalized externalities related to learning, climate change, and stranded capital also contribute to the delayed transition. Intuitively, the capital stock (Panel A) and total consumption (Panel B) of the social planner and the market economy evolve more similarly due to the higher savings rate. As shown in Panel C, the growth rate of consumption in the social optimum after the transition to clean production is 2.26%, about twice as high as in the market economy, where the growth rate of consumption is estimated to be 1.12%. Due to the higher propensity to save, the growth rate of consumption in the market economy is now significantly higher than with the original saving function, where consumption could only grow at 0.35% (compare with Panel C of Figure 9).

6 Extension: Non-convexity in clean production

In this section, we modify the production function of the clean technology to account for the fact that, in addition to the already implemented transition costs, large-scale infrastructure investments are required before the clean technology can become operational. To account for this characteristic, we extend the baseline model by imposing a non-convexity in the production structure of the clean technology. In particular, we assume that the clean production function cannot generate positive output until capital attains a minimum value \bar{k} , leaving the remaining assumptions of the model unchanged.³³ By generating a non-convexity in the production structure of the clean technology, this sort of "set-up cost" specification provides scope for multiple steady states making the characterization of the growth patterns substantially more complex.

As in the baseline model, capital formation with the clean technology entails productivity gains which are not internalized by individual firms. Due to symmetry, the clean production function is given by

$$y_c = A_c \left(k_c - \bar{k} \right)^{\alpha_c + \nu_c},$$

where $\bar{k} > 0$ denotes the minimal capital requirement while ν_c and α_c continue to represent the strength of the learning-by-doing effect and capital intensity, respectively. Firms using the clean technology maximize their profits by choosing labor and capital inputs. Assuming that $\alpha_c + \nu_c = \gamma_c$, we obtain the factor prices summarized in Table 3. As in

Technology:	Rental rate (p) :	Wage (w) :
Dirty technology:	$p_d = \alpha_d A_d k_d^{\gamma_d - 1}$	$w_d = (1 - \alpha_d) A_d k_d^{\gamma_d}$
Clean technology with \bar{k} :	$p_c = \alpha_c A_c (k_c - \bar{k})^{\gamma_c - 1}$	$w_c = (1 - \alpha_c) A_c \left(k_c - \bar{k}\right)^{\gamma_c}$

Table 3: Factor prices: clean technology with capital threshold

the baseline model, the household rents out the capital to the technology that yields the highest return. Since the minimum capital requirement \bar{k} does not affect the investment returns due to the AK-structure in clean production, the tipping points from the baseline economy given by (11) and (12) also remain unchanged.

Unlike the tipping points, capital formation changes due to the minimum capital requirement revealing novel insights regarding the growth patterns of the economy. As above, we set $\gamma_c = 1$, so that capital evolves over time according to

$$k_{t+1} = \begin{cases} \frac{\beta}{1+\beta} \left[(1-\alpha_c)A_c(1-\theta)(k_t-\bar{k}) + (1-\theta-\delta-\eta\varphi(1-\iota))k_t \right] & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t = t^{TP}, \\ \frac{\beta}{1+\beta} \left[(1-\alpha_c)A_c(k_t-\bar{k}) + (1-\delta-\eta\varphi(1-\iota))k_t \right] & \text{if } k_t \ge \bar{k}^{TP} \text{ and } t > t^{TP}, \end{cases}$$

for given $k_0 > 0$ and with t^{TP} referring to the point in time at which the transition dirty to clean occurs.³⁴

 $^{^{33}}$ Our model is similar to the work by Golub & Toman (2016), who also incorporate a minimum capital threshold into the production function of the clean technology. Using numerical simulations, the authors show that a lower capital threshold is associated with a lower probability of an economy falling into an environmental trap.

³⁴Note that the dynamics for ΔP_t and k_{t+1} when $k_t < k^{TP}$ do not change compared to the baseline

Turning to the steady state analysis, the minimum capital threshold and the resulting non-convexity in the production structure gives rise to multiple steady states that differ in terms of their stability properties. Since the steady state under the dirty technology (14) and the tipping point (11) remain unchanged from the baseline framework, the transition condition described by (16) does not change either. However, with a fixed capital requirement in clean production, we find a steady state with a constant level of capital not only for the dirty technology but also for the clean technology. The latter is $unstable^{35}$ and reads

$$k_c^{SS} = \frac{\bar{k}}{1 - \underbrace{\frac{\beta^{-1} + \delta + \eta\varphi(1-\iota)}{(1-\alpha_c)A_c}}_{=\chi}}.$$
(27)

For k_c^{SS} to take a positive value, $\chi < 1$ must hold in (27), which is equivalent to satisfying the green growth condition from the baseline model described by (17).³⁶ If this is not the case, k_c^{SS} does not exist, and thus the capital stock cannot grow under the clean technology. Therefore, meeting the green growth condition is a first prerequisite for generating long-term green growth in the presence of a capital threshold. The second condition that must be satisfied for the clean technology to generate sustained growth is $k_{c,t+1}/k_{c,t} > 1$ or, equivalently,

$$\frac{k_{c,t+1}}{k_{c,t}} = \frac{\beta}{1+\beta} \left[(1-\alpha_c) A_c \left(1 - \frac{\bar{k}}{k_{c,t}} \right) + (1-\delta - \eta \varphi (1-\iota)) \right] > 1.$$
(28)

Unlike in the baseline model, this condition is now increasing in $k_{c,t}$ and equal to unity at k_c^{SS} , which implies that

$$\frac{k_{c,t+1}}{k_{c,t}} \gtrless 1 \quad \Leftrightarrow \quad k_{c,t} \gtrless k_c^{SS}$$

Hence, in presence of a capital threshold, a permanent growth path under the clean technology can only occur if the accumulated stock of capital exceeds the steady state for the clean technology, i.e., $k_{c,t} > k_c^{SS}$. This corresponds to a scale effect induced by the capital threshold, which states that the clean technology can generate long-term growth only beyond a certain size of accumulated capital. Thus, the economy must not only have a high clean factor productivity $(A_c > \bar{A}_c^G)$ to ensure the existence of the clean steady state, but also be sufficiently rich in capital so that the capital stock actually increases with the clean technology. The latter poses an additional barrier to long-term

model.

³⁵It is straightforward to show that $k_{c,t+1} < k_{c,t}$ for $k_{c,t} < k_c^{SS}$ and $k_{c,t+1} > k_{c,t}$ for $k_{c,t} > k_c^{SS}$. ³⁶From $\chi < 1$ in (27), it directly follows that $A_c > (\beta^{-1} + \delta + \eta \varphi(1-\iota)) \frac{1}{1-\alpha_c} = \bar{A}_c^G$.

green growth.

From Equation (27), we can conclude that the clean steady state k_c^{SS} is low and thus sustained green growth is more feasible when both the minimum capital requirement \bar{k} and χ are low. A low χ (and thus a low k_c^{SS}), in turn, is associated with a steep slope of the $k_{c,t+1}$ -function, which will be an important relation for the subsequent graphical analysis.³⁷ Therefore, economies with high values of A_c , ι , and β and low values of \bar{k} , φ and η are, ceteris paribus, more likely to meet $k_{c,t} > k_c^{SS}$ and thus to achieve green growth in the long run.

Providing the analogue to Figure 1 is not trivial, as the dynamics of the economy now depend on the constellation of the three critical values for capital, namely k_d^{SS}, k_c^{SS} and \bar{k}^{TP} . Assuming $\bar{A}_c^G < \bar{A}_c^T$ yields seven different outcomes for the economy and for the opposite case, $\bar{A}_c^G > \bar{A}_c^T$, we arrive at five different outcomes. We discuss all possible constellations and the resulting growth dynamics in the Appendix. For consistency with the previous analysis, we again focus on the case where $\bar{A}_c^G < \bar{A}_c^T$ (which holds under fairly general conditions), and we show in Figures 13 and 14 the graphical representations for a subset of the possible long-run constellations. In particular, we illustrate the following cases: Sustained growth with the clean technology (Case 1), the coexistence of cyclical fluctuations and sustained growth (Case 2), the steady state with the dirty technology (Case 3), and the carbon trap (Case 4).

We illustrate sustained growth with the clean technology in Case 1 of Figure 13. For this growth pattern to occur, the productivity of the green technology must be sufficiently high so that both the transition and the green growth condition are satisfied, i.e., $A_c > \bar{A}_c^T > \bar{A}_c^G$. This, as we have shown above, ensures the existence of the steady state with the clean technology. This clean steady state, k_c^{SS} , must be smaller than k_d^{SS} and \bar{k}^{TP} , which is equivalent to A_c being large. This is fairly intuitive, since the capital threshold exerts pressure on the productivity of the clean technology: To compensate for the need of a sufficiently large capital stock, the clean technology must entail rapid capital formation or, equivalently, a steep capital accumulation function, $k_{c,t+1}$. For all $k_t < \bar{k}^{TP}$, the economy uses the dirty technology until it transitions to the clean one. At this point, due to the steep $k_{c,t+1}$ -function, the economy has surpassed k_c^{SS} so that growth becomes unbounded. Accordingly, sustained growth with the clean technology is the only possible long-run outcome.

If, however, k_c^{SS} is greater than k_d^{SS} and \bar{k}^{TP} (while $A_c > \bar{A}_c^T > \bar{A}_c^G$ still holds), both permanent cyclical fluctuations and sustained growth with the clean technology can emerge, depending on the economy's initial capital endowment. We refer to the Appendix for the proof and illustrate this growth pattern in Case 2 of Figure 13. The key difference between Case 1 and Case 2 is the speed at which the economy can accumulate capital

³⁷Formally, the growth rate of the clean technology with a capital threshold converges to that without a threshold as $k_{c,t}$ increases i.e., $\lim_{k_{c,t}\to\infty} (1 - \bar{k}/k_{c,t}) = 1$.

Case 1: Sustained growth with clean technology

Case 2: Cyclical fluctuations & sustained growth



Figure 13: Phase diagrams with capital threshold \bar{k}

with the clean technology. The lower speed in Case 2 compared to Case 1 is reflected in the flatter slope of the $k_{c,t+1}$ -function, which leads to $k_c^{SS} > \bar{k}^{TP}$ (while $k_c^{SS} < \bar{k}^{TP}$ holds in Case 1). All economies with a capital stock above k_c^{SS} experience unbounded green growth while those with a capital stock below k_c^{SS} fall into permanent cyclical fluctuations. These economies are subject to such fluctuations even though green growth would be technologically feasible, so that with a minimal capital threshold a type of carbon trap can emerge even if clean factor productivity is high.³⁸ This new insight is summarized in the following proposition:

Proposition 7. If the clean technology is subject to a minimal capital threshold and one of the following conditions is fulfilled

- i. capital accumulation with the clean technology proceeds only slowly (i.e., the slope of the $k_{c,t+1}$ -function is relatively flat),
- ii. and/or the minimal capital requirement \bar{k} is relatively high,

³⁸Proposition 7 indicates that sufficiently rapid capital accumulation is required with the clean technology in order to prevent adverse growth effects and/or a return to dirty technology. This finding is in line with Golub & Toman (2016) who conclude that "achieving sufficiently rapid growth in use of the new technology is one key element for avoiding an environmental growth trap." This conclusion can also be applied to the current situation in the European Union, where several member states including France and Germany are now dependent on alternative forms of energy due to the war-induced gas shortage (LeMonde, 2022). In the past, however, not enough has been invested in the infrastructure of renewable technologies, which are therefore unable to close the current energy gap, forcing these member states to return to dirty coal and liquid gas, although these countries have a relatively high total factor productivity. This return not only complicates the achievement of the Paris climate goals but also seems to have negative growth effects on these economies LeMonde (2022).

a capital-poor economy may be trapped in a state of permanent cyclical fluctuations, even if its clean factor productivity is high.

If A_c is below \bar{A}_c^T and \bar{A}_c^G , a unique steady state with the dirty technology arises, which is shown in Case 3 of Figure 14. This growth pattern is not affected by the capital threshold because the productivity of the green technology is so low that the green growth condition is not met. Accordingly, there is no steady state with the clean technology, and the transitional dynamics follow the same pattern as described in Case 2 of Figure 2 of the baseline model.



Figure 14: Phase diagrams with capital threshold \bar{k}

Finally, we turn to the carbon trap which arises if the green growth condition is satisfied but not the transition condition, i.e. $\bar{A}_c^G < A_c < \bar{A}_c^T$. This growth pattern is illustrated in Case 4 of Figure 14. Since the green growth condition is satisfied, there is a steady state with the clean technology and, the capital threshold affects the resulting growth dynamics by influencing the set of capital values from which an economy falls into the carbon trap. The upper trap region is now given by the interval between k_d^{SS} and k_c^{SS} , where its size depends positively on \bar{k} and negatively on the rate at which the economy can accumulate capital with the clean technology: The lower the rate of capital accumulation with the clean technology (i.e., the flatter the slope of the $k_{c,t+1}$ -function), the larger is the size of the upper trap region.

Notably, an economy exposed to high climate-related capital losses not only reaches a lower dirty steady state $(\partial k_d^{SS}/\partial \eta < 0)$, as in the baseline model, but also experiences a simultaneous increase of the clean steady state $(\partial k_c^{SS}/\partial \eta > 0)$ caused by a downward shift of the $k_{c,t+1}$ -function, with both effects enlarging the upper trap region. Moreover, because of the interval between \bar{k}^{TP} and k_c^{SS} , economies revert to k_d^{SS} also after passing the tipping point \bar{k}^{TP} , so that an economy remains stuck in the carbon trap even after the technology transition. This is in stark contrast to our findings from the baseline model; we observe that a capital endowment that ensures the transition to clean production does not necessarily translate into sustained growth anymore. We summarize this finding in the following proposition:

Proposition 8. With a minimal capital requirement in clean production, the economy may be trapped in a low-income, polluting steady state even after the transition to clean production.

Proposition 8 suggests that providing clean technologies to capital-constrained economies may not be enough to overcome the carbon trap. To escape the upper trap region, clean technologies must be complemented by sufficient physical capital, e.g., in the form of a fossil-free infrastructure. The coexistence of cyclical fluctuations and sustained growth (Case 2) as well as the carbon trap (Case 4) provide a theoretically sound foundation for technology-enhancing public policies. A temporary exogenous shock may enable the economy to overcome the critical gap between \bar{k}^{TP} and k_c^{SS} (see Case 2 and 4) and to transition from the inferior dirty steady state to the superior steady state with unlimited green growth.

7 Discussion

So far, we have considered a global economy. However, we can also adopt a country perspective in which case pollution is exogenous. Because climate change damages an economy's productive capacity, countries that are regularly hit by capital-destroying environmental disasters cannot build up sufficient physical capital to reach the level of development where clean technologies are more profitable than dirty ones. Due to unfavorable geographical conditions, low clean factor productivity, and capital endowments, less developed countries (e.g. sub-Saharan countries, small island states or coastal low-lands) are generally less likely to build the capital stock needed for technology transition and are therefore particularly at risk of falling into a carbon trap or a polluting steady state in which long-term growth is not feasible.

A policy recommendation that emerges from the model is to facilitate access to clean and efficient technologies for capital-constrained countries. Beside the climate policies discussed further possible policies are promoting technology and knowledge transfers or granting financial resources (e.g. technology funds) for technology adoption to capitalpoor countries.

Moreover, our model suggests that the choice of technology has a direct impact on a country's long-term growth prospects, with the reason being twofold. First, the technology change enhances capital accumulation through an increase in capital productivity due to stronger spillover effects in clean production. Second, the lower pollution intensity of the clean technology translates into a lower rate of depreciation and thus into higher capital accumulation. Hence, climate policies that support technology transition may serve as a tool to improve development prospects in less developed economies.³⁹

8 Conclusion

This paper develops a tractable two-period overlapping generations model in which capital accumulation, capital depreciation, technology choice and economic growth are endogenous. On the one hand, economies can invest in a conventional dirty technology that is readily available but entails high pollution. On the other hand, economies can adopt a clean technology that causes less pollution — but only after the economy has borne the costs of the transition. We highlight the ambiguous role of capital accumulation for technology transition and long-term economic development: While some capital stock is required for the transition to cleaner technologies, pollution caused by capital accumulation can impede this transition by damaging physical capital.

In this framework, we derive two endogenous tipping points for capital that govern the technology transition from dirty to clean technologies and vice versa, creating a variety of potential long-term outcomes that depend, among others, on the economy's capital endowment, pollution intensity, climate vulnerability, and clean factor productivity. We find that economies with a wide range of clean factor productivities can fall into a carbon trap, defined as a long-term steady state with high pollution and low income. Such a trap emerges in our model even without the interaction of pollution, life expectancy and capital accumulation. As a form of a long-term hidden trap, we also show that an economy can fall into a state of permanent cyclical fluctuations with recurring technological changes.

We show numerically that the social planner switches to clean production three generations (≈ 90 years) earlier than the market economy. Once the savings rates of the household and the social planner are aligned, the socially optimal tipping point occurs one period (≈ 30 years) before the market tipping point, and this time lag is entirely attributable to the non-market effects in the economy. Moreover, we demonstrate theoretically and numerically that a tax levied on dirty production or a subsidy for clean production can implement the socially optimal tipping point in the market economy, reducing the risk of being kept in a perpetual state of low development and high pollution. To overcome the carbon trap, the tax seems to be the preferred instrument in our model framework, as the released public funds provide an additional incentive for capital accumulation, which in turn further reduces the size of the upper trap region.

³⁹We refer to Bretschger & Valente (2011), Bretschger & Suphaphiphat (2014) and Bretschger (2017) for a more detailed discussion on the growth effects of global warming and climate policy.

In an extension of the model, we introduce a non-convexity in the production structure of the clean technology in the form of a minimal threshold for capital, which represents an additional barrier to sustained green growth. To avoid a carbon trap in the presence of a capital threshold, the relevant clean technologies need to be complemented by sufficient physical capital, underlying the need to provide a fossil-free infrastructure to support the technology transition. For example, fossil-free heating systems need to be supported by building insulation, which requires additional capital investments in existing buildings, or electric vehicles need to be complemented by charging stations in addition to their deployment to prevent a carbon trap in the long term.

This research can be further developed in several directions, and we highlight two of them here. First, it might be interesting to consider an open economy to examine the role of international capital flows in reaching the tipping point. An open economy may attract additional foreign capital investments, which can help to accelerate the technology transition; especially in the less developed countries where credit constraints are often seen as a major obstacle to long-term capital formation. Second, a non-renewable resource stock could be included in the model. An increasing resource price according to the Hotelling rule could facilitate the transition to low-carbon technologies and possibly prevent economies from falling into a carbon trap. However, this is left for future research.

References

- Antoci, A., Galeotti, M., & Russu, P. (2011). Poverty trap and global indeterminacy in a growth model with open-access natural resources. *Journal of Economic Theory*, 146(2), 569–591.
- Asano, T., Kunieda, T., & Shibata, A. (2012). Complex behaviour in a piecewise linear dynamic macroeconomic model with endogenous discontinuity. *Journal of Difference Equations and Applications*, 18(11), 1889–1898.
- Asano, T., Shibata, A., Yokoo, M., et al. (2021). Quasi-periodic motions in a polarized overlapping generations model with technology choice (Tech. Rep.). Kyoto University, Institute of Economic Research.
- Asano, T., Shibata, A., Yokoo, M., et al. (2022). *Technology choice, externalities in production, and chaotic middle-income traps* (Tech. Rep.). Kyoto University, Institute of Economic Research.
- Barbier, E. B., & Hochard, J. P. (2019). Poverty-environment traps. Environmental and Resource Economics, 74, 1239–1271.
- Bretschger, L. (2017). Climate policy and economic growth. *Resource and Energy Economics*, 49, 1–15.
- Bretschger, L., & Karydas, C. (2019). Economics of climate change: Introducing the Basic Climate Economic (BCE) model. *Environment and Development Economics*, 24(6), 560–582.
- Bretschger, L., & Komarov, E. (2023). All inclusive climate policy in a growing economy: The role of human health. *Economics Working Paper Series*, 19.
- Bretschger, L., & Suphaphiphat, N. (2014). Effective climate policies in a dynamic north-south model. *European Economic Review*, 69, 59–77.
- Bretschger, L., & Valente, S. (2011). Climate change and uneven development. The Scandinavian Journal of Economics, 113(4), 825–845.
- Chakravorty, U., Moreaux, M., & Tidball, M. (2008). Ordering the extraction of polluting nonrenewable resources. American Economic Review, 98(3), 1128–44.
- Charlier, D., Pommeret, A., & Ricci, F. (2022). A rationale for the right-to-develop climate policy stance?
- Cremers, E. T. (2006). Dynamic efficiency in the two-sector overlapping generations model. *Journal of Economic Dynamics and Control*, 30(11), 1915–1936.

- Cunha-E-Sa, M. A., & Reis, A. B. (2007). The optimal timing of adoption of a green technology. *Environmental and Resource Economics*, 36(1), 35–55.
- Dam, L. (2006). Firm valuation in an environmental overlapping generations model. University of Groningen.
- Damsgaard, E. F. (2012). Exhaustible resources, technology choice and industrialization of developing countries. *Resource and Energy Economics*, 34(3), 271–294.
- Dao, N. T., & Edenhofer, O. (2018). On the fiscal strategies of escaping povertyenvironment traps towards sustainable growth. *Journal of Macroeconomics*, 55, 253– 273.
- Dechezleprêtre, A., Martin, R., & Mohnen, M. (2013). *Knowledge spillovers from clean* and dirty technologies: A patent citation analysis. Grantham Research Institute on Climate Change and the Environment.
- Dietz, S., & Stern, N. (2015). Endogenous growth, convexity of damage and climate risk: how nordhaus' framework supports deep cuts in carbon emissions. *The Economic Journal*, 125(583), 574–620.
- Dugan, A., Prskawetz, A., & Raffin, N. (2022). The environment, life expectancy and growth in overlapping generations models: A survey (Tech. Rep.). ECON WPS-Working Papers in Economic Theory and Policy.
- Edenhofer, O., Hirth, L., Knopf, B., Pahle, M., Schlömer, S., Schmid, E., & Ueckerdt, F. (2013). On the economics of renewable energy sources. *Energy Economics*, 40, S12–S23.
- Fodha, M., & Seegmuller, T. (2014). Environmental quality, public debt and economic development. *Environmental and Resource Economics*, 57(4), 487–504.
- Golub, A., & Toman, M. (2016). Climate change, industrial transformation, and "environmental growth traps". *Environmental and Resource Economics*, 63(2), 249–263.
- Grossman, G. M., & Krueger, A. B. (1995). Economic growth and the environment. *The Quarterly Journal of Economics*, 110(2), 353–377.
- IEA. (2022). World energy investment. World Energy Investment Report.
- Ikefuji, M., & Horii, R. (2007). Wealth heterogeneity and escape from the poverty– environment trap. *Journal of Public Economic Theory*, 9(6), 1041–1068.
- IPCC. (2022). Poverty, livelihoods and sustainable development.

- IPCC. (2023). Climate change 2023: Synthesis Report. Contribution of Working Groups I, II and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Geneva, Switzerland.
- Iwaisako, T. (2002). Technology choice and patterns of growth in an overlapping generations model. *Journal of Macroeconomics*, 24(2), 211–231.
- Karp, L. S., Peri, A., & Rezai, A. (2021). Selfish incentives for climate policy: Empower the young! Available at SSRN 3933987.
- Komen, M. H., Gerking, S., & Folmer, H. (1997). Income and environmental R&D: empirical evidence from oecd countries. *Environment and Development Economics*, 2(4), 505–515.
- Leichenko, R., & Silva, J. A. (2014). Climate change and poverty: vulnerability, impacts, and alleviation strategies. Wiley Interdisciplinary Reviews: Climate Change, 5(4), 539–556.
- LeMonde. (2022). Despite climate commitments, the EU is going back to coal. Retrieved 2022-09-02, from https://www.lemonde.fr/en/economy/article/2022/09/ 02/despite-climate-commitments-the-eu-is-going-back-to-coal_5995594_19 .html
- Lines, M. (2001). The dynamics of capital accumulation in an overlapping generations model. *Structural change and Economic Dynamics*, 12(1), 75–89.
- Mariani, F., Pérez-Barahona, A., & Raffin, N. (2010). Life expectancy and the environment. Journal of Economic Dynamics and Control, 34(4), 798–815.
- Matsuyama, K. (2007). Credit traps and credit cycles. *American Economic Review*, 97(1), 503–516.
- McDonald, A., & Schrattenholzer, L. (2001). Learning rates for energy technologies. Energy policy, 29(4), 255–261.
- McKibben, B. (2012). Global warming's terrifying new math. *Rolling Stone*, 19(7), 2012.
- Narita, D. (2010). Climate policy, technology choice, and multiple equilibria in a developing economy. *Kiel Working Paper*.
- Nelson, D., & Shrimali, G. (2014). Finance mechanisms for lowering the cost of renewable energy in rapidly developing countries. *Climate Policy Initiative*.

- Otto, I. M., Donges, J. F., Cremades, R., Bhowmik, A., Hewitt, R. J., Lucht, W., ... others (2020). Social tipping dynamics for stabilizing earth's climate by 2050. *Proceedings of the National Academy of Sciences*, 117(5), 2354–2365.
- Palivos, T., Varvarigos, D., et al. (2010). Pollution abatement as a source of stabilisation and long-run growth. *Department of Economics discussion paper 2010_13, University* of Macedonia.
- Pommeret, A., & Schubert, K. (2009). Abatement technology adoption under uncertainty. Macroeconomic Dynamics, 13(4), 493–522.
- Romer, P. M. (1986). Increasing returns and long-run growth. Journal of Political Economy, 94(5), 1002–1037.
- Rubin, E. S., Azevedo, I. M., Jaramillo, P., & Yeh, S. (2015). A review of learning rates for electricity supply technologies. *Energy Policy*, 86, 198–218.
- Stokey, N. L. (1998). Are there limits to growth? International Economic Review, 1–31.
- Tahvonen, O. (1995). Dynamics of pollution control when damage is sensitive to the rate of pollution accumulation. *Environmental and Resource Economics*, 5, 9–27.
- Umezuki, Y., & Yokoo, M. (2019). A simple model of growth cycles with technology choice. *Journal of Economic Dynamics and Control*, 100, 164–175.
- Varvarigos, D. (2014). Endogenous longevity and the joint dynamics of pollution and capital accumulation. *Environment and Development Economics*, 19(4), 393–416.
- Xepapadeas, A. (1997). Economic development and environmental pollution: traps and growth. Structural Change and Economic Dynamics, 8(3), 327–350.

Appendix: Basic model

Investment returns

We provide a more detailed derivation of the investment returns used in the main text based on Table 4. Using a standard definition of a return, r_i is defined as

$$r_i(k_{i,t}) = \frac{p_i(k_{i,t}) \cdot k_{i,t}}{k_t}, \quad i \in \{c, d\},$$

where $p_i(k_{i,t})$ corresponds to the rental price of capital and k_t represents the aggregate capital stock regardless of the technology for which it is used.

We start with Case 1 from Table 4, where capital was employed in dirty production in period t-1 and thus $k_{t-1} = k_{d,t-1}$. In period t, the household decides whether to rent the entire capital stock, k_t , to the dirty or to the clean technology. Assuming that the household invests in the dirty technology in two consecutive periods, we obtain $k_t = k_{d,t}$ in period t and the associated return on investment reads

$$r_d(k_{d,t}) = r_d(k_t) = \frac{p_d(k_t) \cdot k_t}{k_t} = p_d(k_t),$$

which corresponds to the rental price of capital derived in Equation (9) in the main text.

Cases	Period $(t-1)$	Period (t)	Return on Investment (r)
Case 1	dirty	dirty	$r_d(k_t) = p_d(k_t)$
Case 2	dirty	clean	$r_c(k_t) = p_c[(1-\theta)k_t] \cdot (1-\theta)$
Case 3	clean	dirty	$r_d(k_t) = p_d(k_t)$
Case 4	clean	clean	$r_c(k_t) = p_c(k_t)$

 Table 4: Returns on investment

Next, we consider the transition from dirty to clean technology (Case 2), which is more involved as the transition costs now affect the return. In particular, as part of the capital stock becomes obsolete, the clean technology can only use the share $(1 - \theta)$ of the total capital stock in production. Accordingly, the rental price of capital for the technology transition from dirty to clean is equal to $p_c[(1 - \theta)k_t]$ and higher than in Case 1 since the marginal product of capital decreases with the stock of capital used in production. Note, however, that the household receives the return on investment only for the capital stock $(1 - \theta)k_t$ that does not become obsolete due to the technology transition. Accordingly, the return on investment associated with the shift in technology from dirty to clean reads

$$r_c(k_{c,t}) = \frac{p_c(k_{c,t}) \cdot k_{c,t}}{k_t},$$

and using $k_c = (1 - \theta)k_t$ yields

$$r_{c}[(1-\theta)k_{t}] = \frac{p_{c}[(1-\theta)k_{t}] \cdot (1-\theta)k_{t}}{k_{t}} = p_{c}[(1-\theta)k_{t}] \cdot (1-\theta),$$

which deviates from its rental price of capital due to the transition costs.

Let us turn to the Cases 3 and 4, in which the economy uses the clean technology in period t-1 and thus $k_{t-1} = k_{c,t-1}$. In period t, the household can invest the entire capital stock either in the dirty technology (Case 3), which leads to $k_t = k_{d,t}$, or in the clean technology (Case 4) which leads to $k_t = k_{c,t}$. Since the transition costs are irrelevant for both investment options, the investment returns do not differ from the rental prices of capital and we have

$$r_i(k_{i,t}) = r_i(k_t) = \frac{p_i(k_t) \cdot k_t}{k_t} = p_i(k_t), \text{ with } i \in \{c, d\}.$$

As a result, regardless of whether the economy transitions from clean to dirty technology or employs the clean technology in two consecutive periods, the household's return to both investments is equal to the corresponding rental price of capital derived in Equation (9) in the main text.

Threshold conditions

Note that $\bar{A}^G < \bar{A}^T$ iff

$$\frac{\beta^{-1} + \delta + \eta\varphi(1-\iota)}{(1-\alpha_c)\alpha_d} < \frac{\beta^{-1} + \delta + \eta\varphi}{(1-\alpha_d)\alpha_c(1-\theta)},$$

and since $\iota \in (0, 1]$, it also holds that

$$\frac{1-\alpha_d}{1-\alpha_c} < \frac{\alpha_d}{\alpha_c(1-\theta)}.$$

This condition is satisfied if α_d and α_c are not too different. It would be violated if α_d were significantly smaller than α_c , so that the capital share in dirty production is so small that the capital loss associated with the technology transition would not matter.

Social planner

The corresponding Lagrange function is

$$\begin{aligned} \mathcal{L} &= \sum_{t=0}^{\infty} \beta_s^t \left(\log(c_t^1) + \beta \log(c_{t+1}^2) \right) \\ &+ \lambda_t \left[A_c k_{c,t}^{\gamma_c} + A_d k_{d,t}^{\gamma_d} - c_t^1 - c_t^2 - k_{t+1} + \left(1 - \frac{\delta - \eta \varphi(1-\iota)}{(1-\theta)^{\kappa}} \right) k_{c,t} + (1-\delta - \eta \varphi) k_{d,t} \right] \\ &+ \zeta_t \left[k_t - \frac{k_{c,t}}{(1-\theta)^{\kappa}} - k_{d,t} \right]. \end{aligned}$$

We define λ_t as the Lagrange multipliers on the aggregate budget constraint and ζ_t as the multiplier on the capital constraint, and obtain the following first-order conditions:

$$\begin{split} \frac{\partial \mathcal{L}}{\partial c_t^1} &= 0 \Leftrightarrow \beta_s^t \left(\frac{1}{c_t^1} - \lambda_t \right) = 0, \\ \frac{\partial \mathcal{L}}{\partial c_{t+1}^2} &= 0 \Leftrightarrow \beta_s^t \left(\frac{\beta}{c_{t+1}^2} - \beta_s \lambda_{t+1} \right) = 0, \\ \frac{\partial \mathcal{L}}{\partial k_{t+1}} &= 0 \Leftrightarrow -\lambda_t + \beta_s \zeta_{t+1} = 0, \\ \frac{\partial \mathcal{L}}{\partial k_{c,t}} &= 0 \Leftrightarrow \lambda_t \left(\gamma_c A_c k_{c,t}^{\gamma_c - 1} + 1 - \frac{\delta - \eta \varphi (1 - \iota)}{(1 - \theta)^{\kappa}} \right) - \frac{\zeta_t}{(1 - \theta)^{\kappa}} = 0, \\ \frac{\partial \mathcal{L}}{\partial k_{d,t}} &= 0 \Leftrightarrow \lambda_t \left(\gamma_d A_d k_{d,t}^{\gamma_d - 1} + 1 - \delta - \eta \varphi \right) - \zeta_t = 0, \end{split}$$

where \mathcal{L} is the Lagrange function. For optimality, any solution must also satisfy the transversality condition which is given by

$$\lim_{t \to \infty} \beta_s^t \zeta_t k_{t+1} = 0.$$

The first three conditions are common, but the last two deserve attention. By simplifying the last two first order conditions we get

$$\frac{1}{c_t^1} = \frac{\beta(1-\theta)^{\kappa}}{c_{t+1}^2} \left(\gamma_c A_c k_{c,t+1}^{\gamma_c-1} + 1 - \frac{\delta - \eta \varphi(1-\iota)}{(1-\theta)^{\kappa}} \right), \quad \text{and}$$
(29)

$$\frac{1}{c_t^1} = \frac{\beta}{c_{t+1}^2} \left(\gamma_d A_d k_{d,t+1}^{\gamma_d - 1} + 1 - \delta - \eta \varphi \right).$$
(30)

From these two expressions it follows that

$$(1-\theta)^{\kappa} \left(\gamma_c A_c k_{c,t+1}^{\gamma_c - 1} + 1 - \frac{\delta - \varphi \eta + \iota \eta \varphi}{(1-\theta)^{\kappa}} \right) = \gamma_d A_d k_{d,t+1}^{\gamma_d - 1} + 1 - \delta - \eta \varphi,$$

which corresponds to Equation (18) in the main text.

Policy analysis

Carbon tax

We denote by τ_d the constant tax rate imposed by the government on dirty production. Under this carbon tax, output with the dirty technology is given by

$$y_{d,t} = (1 - \tau_d) A_d k_{d,t}^{\gamma d}.$$
 (31)

Firms operating with the dirty technology are assumed to maximize their profits in every period by choosing the amount of labor and capital given the carbon tax τ_d levied on output y_d , which yields

$$p_d = (1 - \tau_d) \alpha_d A_d k_d^{\gamma_d - 1}, \quad \text{and} \tag{32}$$

$$w_d = (1 - \tau_d)(1 - \alpha_d) A_d k_d^{\gamma_d}.$$
(33)

The factor prices for the clean technology remain unchanged and are given in Table 1.

Assuming again that the household chooses the investment opportunity whose return is higher than the alternative, we obtain the following tax-adjusted tipping point:

$$\tilde{k}^{TP} = \left(\frac{(1-\tau_d)\alpha_d A_d}{\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}}$$

From that, we can observe that \tilde{k}^{TP} and τ_d are inversely related. Therefore, a higher carbon tax lowers the critical capital stock and thus promotes a switch in favor of the clean technology.

To find the optimal carbon tax τ_d^* , we equate the socially optimal capital stock with the one resulting in the market economy such that $\tilde{k}^{TP} = \bar{k}^{SP}$ or, equivalently,

$$\left(\frac{(1-\tau_d^*)\alpha_d A_d}{\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}} = \left(\frac{\gamma_d A_d}{(1-\theta)A_c - \theta + \eta\varphi\iota}\right)^{\frac{1}{1-\gamma_d}}.$$

Solving for τ_d^* yields

$$\tau_d^* = 1 - \frac{\gamma_d \alpha_c (1-\theta) A_c}{\alpha_d \left((1-\theta) A_c - \theta + \eta \varphi \iota \right)}, \quad \in (0,1),$$

which corresponds to Equation (23) in the main text. In the following, we assume that tax revenues are redistributed to the young or old generation, respectively, so that the government budget remains balanced in any case, and we examine how these redistribution schemes affect the dynamics of the economy.

Carbon tax redistributed to the young household (main text)

Assuming that tax revenues are redistributed to the young generation, we present a more detailed derivation of the transition condition presented in the main text. We begin with the steady state for the dirty technology. From the dynamic equation for k_t , we find

$$\tilde{k}_d^{SS} = \left(\frac{(1 - \alpha_d + \alpha_d \tau_d)A_d}{\beta^{-1} + \delta + \eta\varphi}\right)^{\frac{1}{1 - \gamma_d}}.$$
(34)

For the transition from dirty to clean to occur, we need $\tilde{k}^{TP} < \tilde{k}^{SS}_d$ or, equivalently,

$$\left(\frac{(1-\tau_d)\alpha_d A_d}{\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}} < \left(\frac{(1-\alpha_d+\alpha_d\tau_d)A_d}{\beta^{-1}+\delta+\eta\varphi}\right)^{\frac{1}{1-\gamma_d}}$$

From that, we can solve for A_c , the productivity of the clean technology:

$$A_c > \frac{(1 - \tau_d)\alpha_d(\beta^{-1} + \delta + \eta\varphi)}{(1 - \alpha_d + \alpha_d\tau_d)\alpha_c(1 - \theta)} \equiv \tilde{A}_c^T,$$
(35)

which corresponds to Equation (22) in the main text. As equation (35) shows, a tax levied on dirty production alters the transition condition two times; both of which reduce the productivity threshold from \bar{A}_c^T to \tilde{A}_c^T . The green growth condition remains the same as in the baseline model.

Carbon tax redistributed to the old household

In this section we illustrate the case where tax revenues are refunded as a lump-sum transfer to the contemporary old agents so that the budget of the government is balanced. When operating with the dirty technology, the intertemporal budget constraint and the optimal savings function change to

$$w_{d,t} + (1 - \theta \psi - \Lambda_d)k_t + \frac{\tau_d y_{d,t+1}}{r_{d,t+1}} = c_t^1 + \frac{c_{t+1}^2}{r_{d,t+1}}, \text{ and}$$
$$s_t = \frac{\beta}{1+\beta} \left[w_{d,t} + (1 - \theta \psi - \Lambda_d)k_t \right] - \frac{1}{1+\beta} \frac{\tau_d y_{d,t+1}}{r_{d,t+1}}.$$

Substituting $y_{d,t+1}$ from (31), $p_{d,t}$ from (32), and $w_{d,t}$ from (33) into the optimal saving function and using $p_{d,t+1} = r_{d,t+1}$, we obtain the following dynamics for capital accumulation under the tax policy:

$$k_{t+1} = \begin{cases} \frac{\beta}{1+\beta} \left(\underbrace{(1-\tau_d)(1-\alpha_d)A_dk_t^{\gamma_d}}_{\text{Wage effect (-)}} + (1-\Lambda_d)k_t \right) \left(1 + \underbrace{\frac{\tau_d}{(1+\beta)(1-\tau_d)\alpha_d}}_{\text{Anticipation effect (-)}} \right)^{-1} & \text{if } k_t < \bar{k}^{TP}, \\ \frac{\beta}{1+\beta} \left[(1-\alpha_c)A_c(1-\theta)k_t + (1-\theta-\Lambda_c)k_t \right] & \text{if } k_t \ge \bar{k}^{TP}; \ t = t^{TP}, \\ \frac{\beta}{1+\beta} \left[(1-\alpha_c)A_ck_t + (1-\Lambda_c)k_t \right] & \text{if } k_t \ge \bar{k}^{TP}; \ t > t^{TP}, \end{cases}$$

for given $k_0 > 0$. The dynamic equation for pollution remains the same as in the baseline framework. Unlike before, a refund of tax revenues to the old generation has a clear negative impact on savings and thus reduces capital formation with the dirty technology. This result is due to two effects, which we call the "wage" and "anticipation" effect in the following. The wage effect in the capital accumulation equation states that a firm that uses the dirty technology pays workers a lower wage because of the tax burden, which reduces their disposable income and thus savings and capital formation. The anticipation effect says that young households anticipate that they will have more disposable income tomorrow due to the tax rebate, which reduces their incentive to save at a young age and thus lowers capital accumulation.

Next, we determine the transition condition under the tax policy. Given the redistribution scheme, the steady state for the dirty technology reads

$$\tilde{k}_d^{SS} = \left(\frac{(1-\tau_d)(1-\alpha_d)A_d}{\beta^{-1}+\delta+\varphi\eta + \frac{\tau_d}{\beta(1-\tau_d)\alpha_d}}\right)^{\frac{1}{1-\gamma_d}}$$

An economy can transition to the clean technology if $\tilde{k}^{TP} < k_d^{SS}$ or, equivalently,

$$\left(\frac{(1-\tau_d)\alpha_d A_d}{\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}} < \left(\frac{(1-\tau_d)(1-\alpha_d)A_d}{\beta^{-1}+\delta+\varphi\eta+\frac{\tau_d}{\beta(1-\tau_d)\alpha_d}}\right)^{\frac{1}{1-\gamma_d}},$$

and solving for A_c yields

$$A_c > \frac{\alpha_d \left(\beta^{-1} + \delta + \eta \varphi\right) + \overbrace{\frac{\tau_d}{\beta(1 - \tau_d)}}^{\text{Anticipation effect } (+)}}{(1 - \alpha_d)\alpha_c(1 - \theta)} \equiv \tilde{A}_c^T$$

The anticipation effect raises the productivity threshold above which the economy can transition to the clean technology from \bar{A}_c^T to \tilde{A}_c^T , while the wage effect cancels out and plays no role in the transition condition. Therefore, a narrower range of values for A_c satisfies the transition condition, which reduces the likelihood that economies can transition to clean production on their transition path. The green growth condition

described by \bar{A}_c^G remains unchanged compared to the model without policy intervention.

Figure 15 indicates the new threshold associated with the modified transition condition. A carbon tax, whose proceeds are redistributed to the old generation, narrows



Figure 15: Thresholds and regions with carbon tax

Region 1 (where long-term growth with clean technology is possible) and widens Region 3 (carbon traps). Therefore, such policies reduce the chances for capital-constrained economies, where green growth would be technologically feasible, to switch to clean technologies, thus increasing the risk of remaining in a state of low income and high pollution.

Overall, the considered tax policy reduces the tipping point but also lowers capital accumulation, so that the economy is less likely to be able to build the capital stock needed to reach this (reduced) tipping point. The latter reflects the widespread concern that (overly) stringent climate policies may have detrimental effects on the process of economic development, thereby potentially delaying rather than accelerating technology transition. This illustrates that the implementation of well-intended climate policies in capital-poor economies should take into account potential unintended side effects that may weaken incentives to accumulate capital and thus foster the emergence of carbon traps.

In the following, we show that a carbon tax whose proceeds are redistributed to the old generation affects not only the set of values for A_c under which a carbon trap exists but also the interval for capital endowments that pitch the economy into the carbon trap. While such a tax policy gives households an incentive to adopt the clean technology by lowering the tipping point from \bar{k}^{TP} to \tilde{k}^{TP} , both the wage and anticipation effect work in the opposite direction. In particular, by reducing the household's incentive to save, both effects shift the capital accumulation equation downward from $k_{d,t+1}$ to $\tilde{k}_{d,t+1}$, which lowers the dirty steady state from k_d^{SS} to \tilde{k}_d^{SS} .

We illustrate these effects in Figure 16, where the carbon tax moves the upper trap region from $[k_d^{SS}, \bar{k}^{TP}]$ to $[\tilde{k}_d^{SS}, \tilde{k}^{TP}]$ and consequently the set of capital values from which the economy converges to the carbon trap. While it is unambiguous that the tax shifts the upper trap region to the left, it is not clear whether this region becomes wider or narrower. Therefore, to determine the size of the upper trap region with and without the carbon tax, we employ the parameter values used in the simulation for the case of the carbon trap. In particular, by setting $A_c = 2.75$ and $\tau_d^* = 0.23$, we find that the upper trap region without the tax policy specified by the interval $[k_d^{SS}, \bar{k}^{TP}] = [5.84, 7.54]$ is



Figure 16: Phase diagram: Carbon trap with tax policy

smaller than the upper trap region with the tax policy given by $[\tilde{k}_d^{SS}, \tilde{k}^{TP}] = [0.41, 3.12]$. Overall, a tax policy designed in this way enlarges the interval of the upper trap region from 1.7 to 2.71 and thus increases the set of initial capital endowments from which economies fall into a carbon trap. Hence, as illustrated in Figure 16, the negative effect of the tax on capital accumulation quantitatively outweighs the tax-induced reduction of the tipping point.

Subsidy

As a second policy instrument, we consider a time-invariant subsidy for the clean technology, denoted by τ_c , such that

$$y_{c,t} = (1 + \tau_c) A_c k_{c,t}.$$
 (36)

When maximizing profits, firms incorporate the subsidy into their decision making, leading to the following factor prices:

$$p_c = (1 + \tau_c)\alpha_c A_c,\tag{37}$$

$$w_c = (1 + \tau_c)(1 - \alpha_c)A_c k_c.$$
 (38)

The rental rate of capital and wage for the dirty technology remain as before and are given in Table 1. Equating investment returns yields the following subsidy-adjusted tipping point:

$$\hat{k}^{TP} = \left(\frac{\alpha_d A_d}{(1+\tau_c)\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}}.$$

Intuitively, a subsidy on clean production facilitates technology transition by reducing the tipping point from \bar{k}^{TP} to \hat{k}^{TP} .

To find the optimal subsidy τ_c^* , we equate the optimal capital stock of the social planner with the market tipping point under the subsidy police, i.e. $\hat{k}^{TP} = \bar{k}^{SP}$, which implies

$$\left(\frac{\alpha_d A_d}{(1+\tau_c^*)\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}} = \left(\frac{\gamma_d A_d}{(1-\theta)A_c - \theta + \eta\varphi\iota}\right)^{\frac{1}{1-\gamma_d}}$$

Solving for τ_c^* yields

$$\tau_c^* = \frac{\alpha_d \left((1-\theta)A_c - \theta + \eta \varphi \iota \right)}{\gamma_d \alpha_c A_c (1-\theta)} - 1, \quad \in (0,1),$$

which corresponds to Equation (26) in the main text. In the following, we study the effect of the subsidy depending on which generation has to pay for it.

Subsidy financed by young generation (main text)

Assuming that the subsidy is financed by the young generation, we provide a more detailed derivation of the transition and green growth condition presented in the main text. Let's start with the steady state under the dirty technology, which remains unchanged from the baseline model and can only be reached if the economy does not switch from dirty to clean technology on its transition path, which occurs if $\hat{k}^{TP} < k_d^{SS}$ or, equivalently,

$$\left(\frac{\alpha_d A_d}{(1+\tau_c)\alpha_c A_c(1-\theta)}\right)^{\frac{1}{1-\gamma_d}} < \left(\frac{(1-\alpha_d)A_d}{\beta^{-1}+\delta+\eta\varphi}\right)^{\frac{1}{1-\gamma_d}}$$

From that, we can solve for A_c , which yields

$$A_c > \frac{\alpha_d(\beta^{-1} + \delta + \eta\varphi)}{(1 - \alpha_d)(1 + \tau_c)\alpha_c(1 - \theta)} \equiv \hat{A}_c^T,$$

which refers to Equation (24) in the main text. In light of the subsidy-adjusted transition condition, we can observe that the subsidy lowers the productivity threshold above which the economy can switch to the green technology. Next we turn to the green growth condition. The clean technology can sustain economic growth if $\hat{k}_{c,t+1}/\hat{k}_{c,t} > 1$ or, equivalently,

$$\frac{\hat{k}_{c,t+1}}{\hat{k}_{c,t}} = \frac{\beta}{1+\beta} \left[(1 - \alpha_c - \alpha \tau_c) A_c + (1 - \delta - \eta \varphi (1 - \iota)) \right] > 1.$$

Solving for A_c implies

$$A_c > \left(\frac{1}{\beta} + \delta + \eta \varphi (1-\iota)\right) \frac{1}{(1-\alpha_c - \alpha_c \tau_c)} \equiv \hat{A}_c^G,$$

which corresponds to Equation (25) in the main text. A subsidy policy designed in this way raises the threshold for clean factor productivity above which sustained green growth is feasible.

Subsidy financed by old generation

In this section, we assume that the subsidy is financed by a tax on the savings of the old generation, so that the government maintains a balanced budget. Given that the clean technology is used, the intertemporal budget constraint and the optimal savings function of the household change to

$$w_{c,t} + (1 - \theta\psi - \Lambda_c)k_t - \frac{\tau_c y_{c,t+1}}{r_{c,t+1}} = c_t^1 + \frac{c_{t+1}^2}{r_{c,t+1}}, \text{ and}$$
$$s_t = \frac{\beta}{1+\beta} \left[w_{c,t} + (1 - \theta\psi - \Lambda_c)k_t \right] + \frac{1}{1+\beta} \frac{\tau_c y_{c,t+1}}{r_{c,t+1}}.$$

While the dynamic equation for pollution remains the same as in the baseline model, capital accumulation modifies under the subsidy policy. In particular, substituting $y_{c,t+1}$ from (36), $r_{c,t+1}$ from (37), and $w_{c,t}$ from (38) into the optimal savings function yields

$$k_{t+1} = \begin{cases} \frac{\beta}{1+\beta} \left[(1-\alpha_d) A_d k_t^{\gamma_d} + (1-\Lambda_d) k_t \right] & \text{if } k_t < \bar{k}^{TP}, \\ \frac{\beta}{1+\beta} \left[(1+\tau_c) (1-\alpha_c) A_c k_t (1-\theta) + (1-\theta-\Lambda_c) k_t \right] \left(1 - \frac{\tau_c}{(1+\beta)(1+\tau_c)\alpha_c} \right)^{-1} & \text{if } k_t \ge \bar{k}^{TP}; \ t = t^{TP}, \\ \frac{\beta}{1+\beta} \left(\underbrace{(1+\tau_c)(1-\alpha_c) A_c k_t}_{\text{Wage effect } (+)} + (1-\Lambda_c) k_t \right) \left(1 - \underbrace{\frac{\tau_c}{(1+\beta)(1+\tau_c)\alpha_c}}_{\text{Anticipation effect } (+)} \right)^{-1} & \text{if } k_t \ge \bar{k}^{TP}; \ t > t^{TP}, \end{cases}$$

for given $k_0 > 0$. Subsidizing clean production clearly has a positive effect on capital formation. This is due to the wage and anticipation effect introduced in the previous section, both of which stimulate capital accumulation. Specifically, firms that adopt the clean technology can pay higher wages because of the additional funds they receive, and this increases the young generation's savings, which in turn stimulates capital formation. This corresponds to the wage effect. In addition, young households anticipate that they will have to finance the clean technology in old age. In order to still be able to cover old-age consumption, the young generation compensates for these additional expenses through additional savings, which in turn increases capital formation. This corresponds to the anticipation effect. The subsidy scheme under consideration not only fosters capital accumulation with the clean technology, but also the transition from dirty to clean technology by lowering the tipping point from \bar{k}^{TP} to \hat{k}^{TP} , so that overall an economy is more likely to transition to clean production under the subsidy policy.

Given the redistribution design, we now determine the conditions for technology transition and green growth. The technological transition from dirty to clean requires that $\hat{k}^{TP} < k_d^{SS}$ or, equivalently,

$$A_c > \frac{\alpha_d(\beta^{-1} + \delta + \eta\varphi)}{(1 - \alpha_d)(1 + \tau_c)\alpha_c(1 - \theta)} \equiv \hat{A}_c^T,$$

where $\partial \hat{A}_c^T / \partial \tau_c < 0$. Note that this transition condition is the same as in the case where the subsidy is financed by the young generation, since \hat{k}^{TP} and k_d^{SS} are also the same.

Next we turn to the green growth condition. Given that the subsidy is financed by the old generation, sustained growth with the green technology can be achieved if $\hat{k}_{c,t+1}/\hat{k}_{c,t} > 1$ or, equivalently,

$$\frac{\hat{k}_{c,t+1}}{\hat{k}_{c,t}} = \frac{\frac{\beta}{1+\beta} \left[(1+\tau_c)(1-\alpha_c)A_c + (1-\delta - \eta\varphi(1-\iota)) \right]}{1 - \frac{\tau_c}{(1+\beta)(1+\tau_c)\alpha_c}} > 1,$$

and solving for A_c yields

$$A_{c} > \frac{\beta^{-1} + \delta + \eta \varphi(1 - \iota) - \overbrace{\frac{\tau_{c}}{\beta(1 + \tau_{c})\alpha_{c}}}^{\text{Anticipation effect (-)}}}{\underbrace{\frac{(1 - \alpha_{c})(1 + \tau_{c})}{\text{Wage effect (-)}}} \equiv \hat{A}_{c}^{G}.$$

From that, we can see that the wage and anticipation effect both lower the productivity threshold above which sustained green growth is feasible. Therefore, in the presence of a subsidy policy, economies with lower A_c levels can achieve sustained growth with the clean technology, preventing these economies from converging to the dirty steady state in the long run. Figure 17 indicates the new productivity thresholds associated with the modified transition and green growth condition, respectively. The subsidy narrows Region 2 (non-growing polluting steady state) and extends Region 1 (sustained growth with the clean technology).

The subsidy also influences the set of initial capital endowments under which an economy converges to the carbon trap, as we show in Figure 18. Due to the subsidy's impact on the transition condition, the upper trap region shrinks from $[k_d^{SS}, \bar{k}^{TP}]$ to



Figure 17: Thresholds and regions with subsidy

 $[k_d^{SS}, \hat{k}^{TP}]$. In addition, the positive impact of the subsidy on capital formation stemming from the wage and anticipation effect leads to an increase in the capital accumulation equation under the clean technology from $k_{c,t+1}$ to $\hat{k}_{c,t+1}$. Figure 18 illustrates the case where the economy converges to the polluting steady state in the long run without the subsidy of the clean technology. However, the introduction of the subsidy shifts the capital accumulation equation of the clean technology, $\hat{k}_{c,t+1}$, above the 45-degree line, making the clean technology productive enough to allow for long-run green growth once the capital needed for technology transition has been built up.



Figure 18: Phase diagram: Carbon trap with subsidy

Appendix: Numerical simulation

Welfare

Table 5 demonstrates that welfare in the social optimum is higher than in the market economy for each of the simulated growth patterns in the main text.

Growth Patterns	Social Welfare $[W]$	Market Welfare $[U]$
Sustained green growth	7.9	2.2
Carbon trap	8.2	5
Climate-induced carbon trap	7.4	3
Cyclical fluctuations	9.1	5.2
Modified savings function	7.9	6

Table 5: Welfare: social optimum and market economy

Climate-induced carbon trap

In this section, we demonstrate that a carbon trap can also arise if an economy is highly exposed to the consequences of global warming. As we have seen in the main text, when climate exposure is given by $\eta = 0.3$, a TFP value of $A_c = 3$ leads to sustained growth with the clean technology, as both the transition condition ($\bar{A}_c^T = 3$) and the green growth condition ($\bar{A}_c^G = 2.7$) are satisfied. However, once we increase climate change vulnerability to $\eta = 0.5$, a value of $A_c = 3$ leads to a climate-induced carbon trap which is shown in Figure 19.



Figure 19: Carbon trap due to high climate exposure

To see, economies with $\eta = 5$ and $A_c = 3$ fulfill the green growth condition $(A_c^G = 2.75)$, but not the transition condition $(A_c^T = 3.3)$. Unlike the market economy, the social planner satisfies the transition condition with corresponding values given by $\bar{k}^{TP} = 5.6$, $\bar{k}^{SP} = 1.9$ and $k_d^{SS} = 4.2$. An optimal tax of $\tau_d^* = 0.28$ or subsidy of $\tau_c^* = 0.39$ can induce the socially optimal tipping point in the market economy. The transitional dynamics follow the same reasoning as in Figure 19 in the main text.

Technical details

We begin by outlining the dynamics of the model and some mathematical analysis. The social planner tipping point reads

$$\bar{k}^{SP} = \left(\frac{\gamma_d A_d}{(1-\theta)A_c - \theta + \eta\varphi\iota}\right)^{\frac{1}{1-\gamma_d}}.$$

Intuitively, it means the following: In period t, the economy uses the dirty technology and consumption is given by c_t ; in period t+1 the economy will use the clean technology, if doing so we allow consumption to grow at the same rate as when the economy remains with the dirty technology. Denoting the period of technology change by t = TP, the linear law of motion for consumption reads

$$c_{TP} = \beta \left((1 + A_c)(1 - \theta) - \delta - \eta \varphi(1 - \iota) \right) c_{TP-1}.$$
(39)

Note that this equation describes the dynamics for aggregate per capita consumption, because in the social optimum consumption of both generations is the same, i.e., $c_t^1 = c_t^2$. After the tipping point, we have that

$$c_{TP+1} = \beta \left(1 + A_c - \delta - \eta \varphi (1 - \iota) \right) c_{TP}, \tag{40}$$

which also holds for all subsequent periods. Furthermore, it holds that

$$k_{TP+1} = ((1+A_c)(1-\theta) - \delta - \eta\varphi(1-\iota))k_{TP} - c_{TP}, \text{ but afterwards}$$
$$k_{TP+2} = (1+A_c - \delta - \eta\varphi(1-\iota))k_{TP+1} - c_{TP+1}.$$

The second equation together with (40) constitute a system of two linear difference equations in two variables which can be solved by backwards induction, so that we obtain

$$k_{TP+2} = \beta (1 + A_c - \delta - \eta \varphi (1 - \iota)) k_{TP+1}.$$

This outcome is not surprising as the AK-model does not possess transitional dynamics and k_t and c_t grow at the same rate. Using the transversality condition, we arrive at

$$c_{TP+1} = (1 - \beta)(1 + A_c - \delta - \eta \varphi(1 - \iota))k_{TP+1}$$

We can use this relation in the Euler-equation (29) with $\kappa = 0$ to obtain

$$c_{TP} = \frac{1-\beta}{\beta} k_{TP+1},$$

and therefore

$$k_{TP+1} = \beta((1+A_c)(1-\theta) - \delta - \eta\varphi(1-\iota))k_{TP}$$

This gives us

$$c_{TP} = (1 - \beta)((1 + A_c)(1 - \theta) - \delta - \eta\varphi(1 - \iota))k_{TP}$$

and from (29) with $\kappa = 1$, we have

$$c_{TP-1} = \frac{1-\beta}{\beta} k_{TP}.$$

This means that if assume a value for k_{TP} , we are able to find c_{TP-1} and all previous values of k_t and c_t . We do so in the following way: Given k_{TP} and c_{TP-1} , we can find k_{TP-1} by using the budget-constraint

$$c_{TP-1} + k_{TP} = A_d k_{TP-1}^{\gamma_d} + (1 - \delta - \eta \varphi) k_{TP-1},$$

where we analytically solve for k_{TP-1} using that the left-hand side is given. From this, we find

$$c_{TP-2} = c_{TP-1} (1 + \gamma_d A_d k_{TP-1}^{\gamma_d - 1} - \delta - \eta \varphi)^{-1} / \beta$$

and then k_{TP-2}

$$A_d k_{TP-2}^{\gamma_d} + (1 - \delta - \eta \varphi) k_{TP-2} = c_{TP-2} + k_{TP-1}.$$
(41)

Note, however, that the choice of k_{TP} is not arbitrary. The reason is that \bar{k}^{SP} is given by parameter values, constituting a lower bound k_{TP} , while the upper bound stems from

$$k_{TP}^{max} = \beta \left[A_d (\bar{k}^{SP} - \epsilon)^{\gamma_d} + (1 - \delta - \eta \varphi) (\bar{k}^{SP} - \epsilon) \right],$$

with ϵ small. If k_{TP} lies above this value than it holds than $k_{TP-1} > \bar{k}^{SP}$, and the technology switch occurs earlier.

Grid points and simulation

We create a grid between the points \bar{k}^{SP} and \bar{k}_{TP}^{max} , and for all the grid points we calculate the convergence path in the manner described above and the balanced growth path after the tipping, also as described above, simply relying on the fact that capital and consumption grow at the same constant rate along this path. Hence we have a set of grid points $k_{TP,1}$, $k_{TP,2}$, $k_{TP,3}$, ..., k_{TP}^{max} and their respective predecessors $k_{TP-1,1}$, $k_{TP-2,1}$, $k_{TP-3,1}$ and so on. When we want to simulate the dynamics for an economy with initial stock of capital k_0 it might be that there is no value $k_{TP-j,i}$ equal to k_0 , where j reflects the time distance to the tipping point and i is the number of the grid point. Hence, we had to write the code such that it can handle any possible values of the economy's initial endowment k_0 , and to allow for this feature we wrote the following procedure: First, take some arbitrary k_0 . Matlab finds the value of k_t in the set of convergences that is closest to k_0 and calculates the subsequent values of k_1 , k_2 and so on as the linear interpolation of the sequences that bracket k_0 . To see, assume that we have some value $k_{TP-j,i}^*$ that is smaller but close to k_0 , so that k_0 is bracketed by $k_{TP-i,i}^*$ and $k_{TP-i,i+1}^*$. For these two values we have already calculated the sequence of capital, i.e. $k_{TP-j+1,i}^*, k_{TP-j+2,i}^*, \dots$, and $k_{TP-j+1,i+1}^*, k_{TP-j+2,i+1}^*$ and so on. Thus, we can calculate the capital stocks that follow k_0 , for instance k_1 , as an linearly interpolated value between $k_{TP-j+1,i}^*$ and $k_{TP-j+1,i+1}^*$.

Appendix: Extension

Minimum capital threshold

In this section, we discuss all the possible configurations and growth patterns when we introduce a minimum capital threshold for the clean technology. First, we begin with the threshold order of $\bar{A}_c^G < \bar{A}_c^T$. We assume that if the capital stock becomes sufficiently small under the clean technology, the economy switches to the dirty technology. Recall that to the left of k_c^{SS} , due to its instability, the capital stock decreases, while to the right the opposite is true.

- 1. $\bar{A}_c^G < \bar{A}_c^T < A_c$: This constellation implies $\bar{k}^{TP} < k_d^{SS}$ and we can distinguish the following cases:
 - (a) $k_c^{SS} < \bar{k}^{TP} < k_d^{SS}$: For all $k_t < \bar{k}^{TP}$, the economy utilizes the dirty technology until it switches to the clean one. At this point, the economy has surpassed the point k_c^{SS} so that growth becomes unbounded. Accordingly, sustained

growth with the clean technology is the only possible long-run outcome. This constellation is represented in Case 1 of Figure 13 in the main text.

- (b) $\bar{k}^{TP} < k_c^{SS} < k_d^{SS}$: For all $k_t < \bar{k}^{TP}$, the economy employs the dirty technology and accumulates capital until the tipping point is reached where the transition to the clean technology occurs. Note, however, that the capital stock is contracting with the clean technology over the course of time until the economy passes below \underline{k}^{TP} and switches back to the dirty technology; a cycle is established. For all $k_t > k_c^{SS}$, the economy utilizes the clean technology and unbounded growth with that technology arises. Thus, overall, sustained growth with the clean technology as well as permanent cyclical fluctuations are possible long-run outcomes. Which one is realized depends on whether the economy's capital initial endowment exceeds k_c^{SS} .
- (c) $\bar{k}^{TP} < k_d^{SS} < k_c^{SS}$: The argumentation is the same as in the previous case. This constellation is illustrated in Case 3 of Figure 14 in the main text.
- 2. $\bar{A}_c^G < A_c < \bar{A}_c^T$: This constellation implies $k_d^{SS} < \bar{k}^{TP}$ and we can distinguish the following cases:
 - (a) $k_c^{SS} < k_d^{SS} < \bar{k}^{TP}$: For all $k_t < \bar{k}^{TP}$ the economy utilizes the dirty technology and converges monotonously to the respective steady state. In the opposite case, the economy relies on the clean technology and sustained growth is feasible, since the economy has passed the point k_c^{SS} . The steady state with the dirty technology and sustained growth with the clean technology are possible long-run outcomes. Which one is realized depends on whether the initial endowment of the economy exceeds \bar{k}^{TP} .
 - (b) $k_d^{SS} < k_c^{SS} < \bar{k}^{TP}$: The argumentation is the same as in the previous case.
 - (c) $k_d^{SS} < \bar{k}^{TP} < k_c^{SS}$: For all $k_t < \bar{k}^{TP}$ the economy employs the dirty technology and converges to the dirty steady state. For all k_t that lie in between of \bar{k}^{TP} and k_c^{SS} the economy temporarily employs the clean technology until it switches back to the dirty one and only capital values that exceed k_c^{SS} lead to long-run growth. The steady state with the dirty technology and sustained growth with the clean technology are possible long-run outcomes. Which one is realized depends now on whether the initial endowment of the economy exceeds k_c^{SS} . This corresponds to the carbon trap illustrated in Case 4 of Figure 14 in the main text.
- 3. $A_c < \bar{A}_c^G < \bar{A}_c^T$: This case implies $k_d^{SS} < \bar{k}^{TP}$ and there is no positive value for k_c^{SS} . Hence, this case is the same as in the model without a minimal capital threshold.

The only possible long-run outcome is a steady state with the dirty technology, which is illustrated in Case 2 of Figure 13 in the main text.

The alternative threshold order, $\bar{A}_c^T < \bar{A}_c^G$, is somewhat simpler. As long as $A_c < \bar{A}_c^G$ there is no positive value for k_c^{SS} and the model behaves as in the case without the capital threshold \bar{k} . If, however, $\bar{A}_c^G < A_c$, the economy behaves as in the case when $\bar{A}_c^G < \bar{A}_c^T$, i.e., Case 1 from above. The relative size of \bar{A}_c^T and \bar{A}_c^G is irrelevant and, in total, five constellations arise.

Working Papers of the Center of Economic Research at ETH Zurich

- (PDF-files of the Working Papers can be downloaded at www.cer.ethz.ch/research/working-papers.html).
- 23/386 L. Bretschger, E. Komarov, M. Leuthard Overcoming the Carbon Trap: Climate Policy and Technology Tipping
- 23/385 M. Alsina-Pujols, I. Hovdahl Patent Protection and the Transition to Clean Technology
- 23/384 L. Bretschger, E. Komarov All Inclusive Climate Policy in a Growing Economy: The Role of Human Health
- 23/383 D. Bounie, A. Dubus, P. Waelbroeck Competition Between Strategic Data Intermediaries with Implications for Merger Policy
- 23/382 J. Lehtomaa, C. Renoir The Economic Impact of Tropical Cyclones: Case Studies in General Equilibrium
- 23/381 S. Srinivasan Social Policies and Adaptation to Extreme Weather: Evidence from South Africa
- 23/380 L. Barrage Fiscal Costs of Climate Change in the United States
- 23/379 S. Bansal, M. Filippini, S. Srinivasan How Regulation Might Fail to Reduce Energy Consumption While Still Stimulating Total Factor Productivity Growth
- 22/378 A. Jo, C. Karydas Firm Heterogeneity, Industry Dynamics and Climate Policy
- 22/377 V. da Cruz Cap-and-Innovate: Evidence of regulation-induced innovation in California
- 22/376 G. Loumeau Land Consolidation Reforms: A Natural Experiment on the Economic and Political Effects of Agricultural Mechanization
- 22/375 L. Bretschger Green Road is Open: Economic Pathway with a Carbon Price Escalator
- 22/374 A. Goussebaïle Democratic Climate Policies with Overlapping Generations

- 22/373 H. Gersbach, O. Tejada, J. Wagner Policy Reforms and the Amount of Checks & Balances
- 22/372 S. Houde, W. Wang The Incidence of the U.S.-China Solar Trade War
- 22/371 J. A. Bingler Expect the worst, hope for the best: The valuation of climate risks and opportunities in sovereign bonds
- 22/370 A. Bommier, A. Fabre, A. GoussebaÃ⁻le, and D. Heyen Disagreement Aversion
- 22/369 A. Jo, A. Miftakhova How Constant is Constant Elasticity of Substitution? Endogenous Substitution between Clean and Dirty Energy
- 22/368 N. Boogen, M. Filippini, A. L. Martinez-Cruz Value of co-benefits from energy saving ventilation systems-Contingent valuations on Swiss home owners
- 22/367 D. Bounie, A. Dubus, P. Waelbroeck Market for Information and Selling Mechanisms
- 22/366 N. Kumar, N. Kumar Raut, S. Srinivasan Herd behavior in the choice of motorcycles: Evidence from Nepal
- 21/365 E. Komarov Capital Flows and Endogenous Growth
- 21/364 L. Bretschger, A. Jo Complementarity between labor and energy: A firm-level analysis
- 21/363 J. A. Bingler, C. Colesanti Senni, P. Monnin Climate Transition Risk Metrics: Understanding Convergence and Divergence across Firms and Providers
- 21/362 S. Rausch, H. Yonezawa Green Technology Policies versus Carbon Pricing: An Intergenerational Perspective
- 21/361 F. Landis, G. Fredriksson, S. Rausch Between- and Within-Country Distributional Impacts from Harmonizing Carbon Prices in the EU
- 21/360 O. Kalsbach, S. Rausch Pricing Carbon in a Multi-Sector Economy with Social Discounting
- 21/359 S. Houde, T. Wekhof The Narrative of the Energy Efficiency Gap

- 21/358 F. Böser, H. Gersbach Leverage Constraints and Bank Monitoring: Bank Regulation versus Monetary Policy
- 21/357 F. Böser Monetary Policy under Subjective Beliefs of Banks: Optimal Central Bank Collateral Requirements
- 21/356 D. Cerruti, M. Filippini Speed limits and vehicle accidents in built-up areas: The impact of 30 km/h zones
- 21/355 A. Miftakhova, C. Renoir Economic Growth and Equity in Anticipation of Climate Policy
- 21/354 F. Böser, C. Colesanti Senni CAROs: Climate Risk-Adjusted Refinancing Operations
- 21/353 M. Filippini, N. Kumar, S. Srinivasan Behavioral Anomalies and Fuel Efficiency: Evidence from Motorcycles in Nepal
- 21/352 V. Angst, C. Colesanti Senni, M. Maibach, M. Peter, N. Reidt, R. van Nieuwkoop Economic impacts of decarbonizing the Swiss passenger transport sector
- 21/351 N. Reidt Climate Policies and Labor Markets in Developing Countries
- 21/350 V. Britz, H. Gersbach Pendular Voting
- 21/349 E. Grieg Public opinion and special interests in American environmental politics
- 21/348 N. Ritter, J. A. BinglerDo homo sapiens know their prices? Insights on dysfunctional price mechanisms from a large field experiment
- 20/347 C. Daminato, M. Filippini, F. Haufler Personalized Digital Information and Tax-favoured Retirement Savings: Quasi-experimental Evidence from Administrative Data